UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported) August 17, 2018

Farmers & Merchants Bancorp, Inc.

(Exact Name of Registrant as Specified in its Charter)

Ohio (State or Other Jurisdiction of Incorporation) 001-38084 (Commission File Number) 34-1469491 (IRS Employer Identification No.)

307 North Defiance Street, Archbold, Ohio (Address of Principal Executive Offices)

43502 (Zip Code)

Registrant's Telephone Number, including Area Code (419) 446-2501

\$N/A\$ (Former Name or Former Address, if Changed Since Last Report)

	Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the ollowing provisions (see General Instruction A.2. below):		
X	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)		
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)		
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17-CFR 240.13e-4(c))		
Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter)			
Emerging growth company \Box			

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01. Entry Into a Material Definitive Agreement.

On August 17, 2018, Farmers & Merchants Bancorp, Inc., an Ohio corporation ("F&M"), and Limberlost Bancshares, Inc., an Indiana corporation ("LBI"), entered into an Agreement and Plan of Reorganization and Merger (the "Merger Agreement"), pursuant to which LBI will, subject to the terms and conditions of the Merger Agreement, merge with and into F&M (the "Merger"), whereupon the separate corporate existence of LBI will cease and F&M will survive. Immediately following the Merger, LBI's wholly-owned subsidiary, Bank of Geneva, will be merged with and into F&M's wholly-owned subsidiary, Farmers & Merchants State Bank, with Farmers & Merchants State Bank as the surviving bank.

Based on the closing price of F&M's common stock on August 17, 2018 of \$43.88 per share, the transaction value for the shares of common stock and cash to be paid is approximately \$88.765 million.

The transaction is expected to be a tax-free stock exchange for LBI's shareholders who will be receiving F&M's common stock pursuant to the Merger, other than as to the 10% of the merger consideration to be paid in cash. Subject to LBI's shareholders' approval of the Merger, regulatory approvals and other customary closing conditions, the parties anticipate completing the Merger in the fourth quarter of 2018 or first quarter of 2019. A copy of the Merger Agreement is filed as Exhibit 1.01 and incorporated herein by reference.

The Boards of Directors of both F&M and LBI have approved the Merger Agreement. The members of the Board of Directors of LBI have entered into a Voting Agreement pursuant to which each of them has agreed to vote their shares of LBI common stock in favor of the Merger. A copy of the form of Voting Agreement is attached to the Merger Agreement as Exhibit B.

Subject to the terms and conditions of the Merger Agreement, upon the completion of the Merger, each share of outstanding LBI common stock, \$10.00 par value, will be converted into: (i) 1,830 shares (the "Exchange Ratio") of F&M common stock, without par value; plus (ii) \$8,465.00 in cash. The Exchange Ratio is subject to adjustments for stock splits, stock dividends, recapitalization, or similar transactions, or as otherwise described in the Merger Agreement. Fractional shares of F&M common stock will not be issued in respect of fractional interests arising from the Exchange Ratio but will be paid in cash pursuant to the Merger Agreement.

The Merger Agreement contains representations, warranties and covenants of LBI and F&M, including, among others, covenants (i) to conduct their respective businesses in the ordinary course during the period between the execution of the Merger Agreement and consummation of the Merger and (ii) prohibiting LBI from engaging in certain kinds of transactions during such period. The Board of Directors of LBI has adopted a resolution recommending the approval and adoption of the Merger Agreement by its shareholders, and LBI has agreed to hold a shareholder meeting to put the Merger before its shareholders for consideration. LBI has also agreed, subject to certain exceptions as set forth in the Merger Agreement, not to (i) solicit proposals relating to alternative business combination transactions or (ii) enter into discussions or negotiations or provide confidential information in connection with any proposals for alternative business combination transactions.

Consummation of the Merger is subject to various conditions, including, among others, (i) requisite approval of the holders of LBI common stock; (ii) effectiveness of a Form S-4 registration statement relating to the F&M common stock to be issued in the Merger and listing of the F&M common stock to be issued in the Merger on the NASDAQ Capital Market; and (iii) receipt of regulatory approvals.

The Merger Agreement contains certain termination rights for both F&M and LBI. Under certain circumstances, termination of the Merger Agreement may result in the payment of a termination fee of \$3,500,000 to F&M, all as more fully described in the Merger Agreement.

The foregoing description of the Merger Agreement is not complete and is qualified in its entirety by reference to the Merger Agreement, which is filed as Exhibit 1.01, and, as mentioned above, is incorporated by reference.

The Merger Agreement also provides that, at the Effective Time, the F&M board of directors will appoint Andrew J. Briggs, Chairman of LBI and President of Bank of Geneva, to the boards of F&M and Farmers & Merchants State Bank.

Item 8.01. Other Events.

On August 20, 2018, F&M issued a press release announcing the execution of the Merger Agreement. A copy of the press release is attached hereto as Exhibit 99.1 and incorporated by reference herein.

FORWARD-LOOKING STATEMENTS

This filing and the exhibits hereto contain forward-looking statements made pursuant to the safe-harbor provisions of the Private Securities Litigation Reform Act of 1995. Such forward-looking statements can often, but not always, be identified by the use of words like "believe", "continue", "pattern", "estimate", "project", "intend", "anticipate", "expect" and similar expressions or future or conditional verbs such as "will", would", "should", "could", "might", "can", "may", or similar expressions. These forward-looking statements include, but are not limited to, statements relating to the expected timing and benefits of the proposed Merger, including future financial and operating results, cost savings, enhanced revenues, and accretion/dilution to reported earnings that may be realized from the Merger, as well as other statements of expectations regarding the Merger, and other statements of F&M's goals, intentions and expectations; statements regarding F&M's business plan and growth strategies; statements regarding the asset quality of F&M's loan and investment portfolios; and estimates of F&M's risks and future costs and benefits, whether with respect to the Merger or otherwise.

These forward-looking statements are subject to significant risks, assumptions and uncertainties that may cause results to differ materially from those set forth in forward-looking statements, including, among other things: the risk that the businesses of F&M and LBI will not be integrated successfully or such integration may be more difficult, time-consuming or costly than expected; anticipated revenue synergies and cost savings from the Merger may not be fully realized or realized within the expected time frame; revenues following the Merger may be lower than expected; customer and employee relationships and business operations may be disrupted by the Merger; the ability to obtain required regulatory and shareholder approval, and the ability to complete the Merger on the expected timeframe; possible changes in economic and business conditions; the existence or exacerbation of general geopolitical instability and uncertainty; possible changes in monetary and fiscal policies, and laws and regulations; the effects of easing restrictions on participants in the financial services industry; the cost and other effects of legal and administrative cases; possible changes in the creditworthiness of customers and the possible impairment of collectability of loans; fluctuations in market rates of interest; competitive factors in the banking industry; changes in the banking legislation or regulatory requirements of federal and state agencies applicable to bank holding companies and banks like Farmers & Merchants State Bank; continued availability of earnings and excess capital sufficient for the lawful and prudent declaration of dividends; changes in market, economic, operational, liquidity, credit and interest rate risks associated with F&M's business; and other risks and factors identified in F&M's filings with the Securities and Exchange Commission.

F&M does not undertake any obligation to update any forward-looking statement, whether written or oral, relating to the matters discussed in this filing. In addition, F&M's and LBI's past results of operations do not necessarily indicate either of their anticipated future results, whether the Merger is effectuated or not.

ADDITIONAL INFORMATION FOR LBI SHAREHOLDERS

This filing does not constitute an offer to sell or the solicitation of an offer to buy any securities or a solicitation of any proxy vote or approval. The proposed Merger will be submitted to LBI's shareholders for their consideration. In connection with the proposed Merger, F&M will file with the SEC a Registration Statement on Form S-4 that will include a Proxy Statement for LBI and a Prospectus of F&M, as well as other relevant documents concerning the proposed transaction. SHAREHOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND THE CORRESPONDING PROXY STATEMENT-PROSPECTUS REGARDING THE MERGER WHEN THEY BECOME AVAILABLE, AS WELL AS ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, TOGETHER WITH ALL AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, AS THEY WILL CONTAIN IMPORTANT INFORMATION.

Once filed, you may obtain a free copy of the Proxy Statement – Prospectus, when it becomes available, as well as other filings containing information about F&M and LBI, at the SEC's website (http://www.sec.gov). You may also obtain these documents, free of charge, by accessing F&M's website (http://www.fm.bank) under the tab "About Us", then to the heading "Investor Relations," and finally under the link "SEC Filings and Documents".

LBI and its directors and executive officers may be deemed to be participants in the solicitation of proxies from the shareholders of LBI in connection with the proposed Merger. Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the Proxy Statement – Prospectus regarding the proposed Merger when it becomes available. Free copies of this document may be obtained as described in the preceding paragraph.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	Description of Exhibit	
1.01	Agreement and Plan of Reorganization and Merger between Farmers & Merchants Bancorp, Inc. and Limberlost Bancshares, Inc. dated August 17, 2018.	
99.1	Company Press Release dated August 20, 2018.	

Signatures

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto authorized.

Dated: August 20, 2018

FARMERS & MERCHANTS BANCORP, INC.

(Registrant)

/s/ Paul S. Siebenmorgen

Paul S. Siebenmorgen

President & Chief Executive Officer

/s/ Barbara J. Britenriker

Barbara J. Britenriker Executive Vice President & Chief Financial Officer

Exhibit 1.01

Agreement and Plan of Reorganization and Merger between Farmers & Merchants Bancorp, Inc. and Limberlost Bancshares, Inc. dated August 17, 2018.

AGREEMENT AND PLAN OF REORGANIZATION AND MERGER

BETWEEN

FARMERS & MERCHANTS BANCORP, INC.

AND

LIMBERLOST BANCSHARES, INC.

Dated as of August 17, 2018

TABLE OF CONTENTS

CECTION 4	THE MED CED C	Page
SECTION 1	THE MERGERS	1
1.1 1.2	LBI Merger	1 2
1.3	The Bank Merger Right to Revise Mergers	2
1.5	right to revise mergers	2
SECTION 2	EFFECT OF THE MERGER	2
2.1	General Description	2
2.2	Name, Offices, and Management	2
2.3	Capital Structure	2
2.4	Articles of Incorporation and Code of Regulations	2
2.5	Assets and Liabilities	3
2.6	Additional Actions	3
SECTION 3	CONSIDERATION TO BE DISTRIBUTED	3
3.1	Consideration	3
3.2	Fractional F&M Common Shares	3
3.3	Recapitalization	3
3.4	Distribution of F&M's Common Stock	4
SECTION 4	DISSENTING SHAREHOLDERS	5
SECTION 5	REPRESENTATIONS AND WARRANTIES OF LBI	ϵ
5.1	Organization and Authority	6
5.2	Authorization	6
5.3	Capitalization	8
5.4	Organization Documents	g
5.5	Compliance with Law	S
5.6	Accuracy of Statements	S
5.7	Litigation and Pending Proceedings	10
5.8	Financial Statements	10
5.9	Absence of Certain Changes	10
5.10	Absence of Undisclosed Liabilities	11
5.11	Title to Assets	11
5.12	Loans and Investments	11
5.13	Employee Benefit Plans	12
5.14	Obligations to Employees	15
5.15	Taxes, Returns and Reports	16
5.16	Deposit Insurance	16
5.17	Reports	16
5.18	Absence of Defaults	17
5.19	Tax and Regulatory Matters	17
5.20	Real Property	17
5.21	Securities Law Compliance	19
5.22	Broker's or Finder's Fees	19

5.23	Shareholder Rights Plan	19
5.24	Indemnification Agreements	19
5.25	Agreements with Regulatory Agencies	19
5.26	Nonsurvival of Representations and Warranties	19
SECTION 6	REPRESENTATIONS AND WARRANTIES OF F&M	19
6.1	Organization and Qualification	20
6.2	Authorization	20
6.3	Capitalization	21
6.4	Organizational Documents	21
6.5	Compliance with Law	21
6.6	Accuracy of Statements	22
6.7	Litigation and Pending Proceedings	22
6.8	Financial Statements	22
6.9	Absence of Certain Changes	23
6.10	Taxes, Returns and Reports	23
6.11	Deposit Insurance	24
6.12	Reports	24
6.13	Absence of Defaults	24
6.14	Tax and Regulatory Matters	24
6.15	Securities Law Compliance	24
6.16	Broker's or Finder's Fees	25
6.17	Agreements with Regulatory Agencies	25
6.18	Sufficient Funds	25
6.19	No Shareholder Approval	25
6.20	Nonsurvival of Representations and Warranties	25
SECTION 7	COVENANTS OF LBI	25
7.1	Shareholder Approval	25
7.2	Other Approvals	26
7.3	Conduct of Business	26
7.4	Preservation of Business	29
7.5	Other Negotiations	29
7.6	Announcement; Press Releases	30
7.7	LBI Disclosure Letter	30
7.8	Confidentiality	30
7.9	Cooperation	31
7.10	LBI Fairness Opinion	31
7.11	Financial Statements and Other Reports	31
7.12	Adverse Actions	31
7.13	Bank Merger Agreement	31
7.14	Employment Agreement and Change in Control Agreement	31
7.15	ESOP and 401(k) Profit Sharing Plan	32
SECTION 8	COVENANTS OF F&M	32
8.1	Approvals	32
8.2	Employee Benefit Plans	32

8.3	Announcement; Press Releases	34
8.4	Confidentiality	34
8.5	Directors and Officers Insurance	34
8.6	SEC and Other Reports	36
8.7	F&M Disclosure Letter	36
8.8	Adverse Actions	36
8.9	Cooperation	36
8.10	Bank Merger Agreement	36
8.11	Preservation of Business	36
8.12	Representation on F&M and F&M Bank Boards	37
SECTION 9	CONDITIONS PRECEDENT TO THE MERGER AND THE BANK MERGER	37
9.1	Shareholder Approval	37
9.2	Registration Statement Effective	37
9.3	Tax Opinions	37
9.4	Regulatory Approvals	37
9.5	Officer's Certificate	38
9.6	Secretary's Certificate	38
9.7	No Judicial Prohibition	38
9.8	LBI Fairness Opinion	38
9.9	Bank Merger Agreement	38
9.10	Employment Agreement	38
9.11	Appointment to Board of Directors	38
9.12	Exchange Fund	39
9.13	D & O Tail Insurance	39
9.14	Miscellaneous	39
SECTION 10	TERMINATION OF MERGER	39
10.1	Manner of Termination	39
10.2	Effect of Termination	40
SECTION 11	EFFECTIVE DATE OF MERGER	41
SECTION 12	CLOSING	41
12.1	Closing Date and Place	41
12.2	Merger - Certificate of Merger and Articles of Merger	41
12.3	Bank Merger - Certificate of Merger and Articles of Merger	41
SECTION 13	MISCELLANEOUS	41
13.1	Effective Agreement	41
13.2	Waiver; Amendment	42
13.3	Notices	42
13.4	Headings	43
13.5	Severability	43
13.6	Counterparts	43
13.7	Governing Law	43

Entire Agreement	43
Expenses	43
Securityholder Litigation	43
Certain Definitions	44
Survival of Contents	44
	Expenses Securityholder Litigation Certain Definitions

Exhibits Exhibit A- Bank Merger Agreement Exhibit B- Voting Agreement

AGREEMENT AND PLAN OF REORGANIZATION AND MERGER **BETWEEN** FARMERS & MERCHANTS BANCORP, INC. AND

LIMBERLOST BANCSHARES, INC.

THIS AGREEMENT AND PLAN OF REORGANIZATION AND MERGER (the "Agreement"), is entered as of the 17th day of August, 2018, by and between FARMERS & MERCHANTS BANCORP, INC., an Ohio corporation ("F&M") and LIMBERLOST BANCSHARES, INC., an Indiana corporation ("LBI").

WITNESSETH:

WHEREAS, F&M is a registered financial holding company under the Bank Holding Company Act of 1956, as amended, with its principal place of business in Archbold, Ohio, with Farmers & Merchants State Bank, an Ohio commercial bank ("F&M Bank") as its wholly-owned subsidiary;

WHEREAS, LBI is a registered bank holding company under the Bank Holding Company Act of 1956, as amended, with its principal place of business in Geneva, Indiana, with Bank of Geneva, an Indiana commercial bank (the "Bank") as its wholly-owned subsidiary;

WHEREAS, it is the desire of F&M and LBI to effect a series of transactions whereby (i) LBI will merge with and into F&M (the "Merger"), and (ii) the Bank will merge with and into F&M Bank;

WHEREAS, the Boards of Directors of F&M and LBI have approved this Agreement and authorized its execution; and

WHEREAS, for federal income tax purposes, it is intended that the Merger shall qualify as a "reorganization" within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the "Code"), and F&M and LBI desire to and hereby adopt this Agreement as a plan of reorganization for purposes of Sections 354 and 361 of the Code;

NOW, THEREFORE, in consideration of the mutual promises, covenants, and agreements herein contained and other good and valuable consideration, the receipt of which is hereby acknowledged, F&M and LBI hereby make this Agreement and prescribe the terms and conditions of the merger of LBI with and into F&M and the Bank with and into F&M Bank and the mode of carrying the transactions into effect as follows:

SECTION 1

THE MERGERS

1.1 LBI Merger. Subject to the terms and conditions of this Agreement, on the Effective Date (as defined in Section 11 hereof), LBI shall be merged with and into F&M, which shall be the "Continuing Company" and shall continue its corporate existence under the laws of the State of Ohio, pursuant to the provisions of and with the effect provided in the Ohio General Corporation Law ("OGCL").

- 1.2 <u>The Bank Merger</u>. Subject to the terms and conditions of this Agreement, on the Effective Date and immediately after the Merger, the Bank shall be merged with and into F&M Bank pursuant to the terms and conditions of the Agreement and Plan of Merger attached hereto as <u>Exhibit A</u> (the "Bank Merger Agreement") and otherwise in accordance with 12 U.S.C. §1828(c) and the Ohio banking law, Chapter 11 of the Ohio Revised Code, as amended, together with any regulations promulgated thereunder (the "Bank Merger").
- 1.3 Right to Revise Mergers. The parties may, at any time, change the method of effecting the Merger or the Bank Merger if and to the extent the parties deem such change to be desirable, including, without limitation, to provide for the merger of LBI into a wholly-owned subsidiary of F&M and/or the merger of the Bank into wholly-owned subsidiaries of F&M or F&M Bank; provided, however, that no such change, modification or amendment shall (a) alter or change the amount or kind of consideration to be received by the shareholders of LBI specified in Section 3 hereof as a result of the Merger, except in accordance with the terms of Section 3 hereof; (b) adversely affect the tax treatment to the shareholders of LBI; or (c) materially impede or delay receipt of any approvals referred to in this Agreement or the consummation of the transactions contemplated by this Agreement.

SECTION 2

EFFECT OF THE MERGER

Upon the Merger becoming effective:

- 2.1 <u>General Description</u>. The separate existence of LBI shall cease and the Continuing Company shall possess all of the assets of LBI and all of its rights, privileges, immunities, powers, and franchises and shall be subject to and assume all of the duties, obligations and liabilities of LBI.
- 2.2 <u>Name, Offices, and Management</u>. The name of the Continuing Company shall continue to be "Farmers & Merchants Bancorp, Inc." Its principal office shall be located at 307 North Defiance Street, Archbold, Ohio 43502. The Board of Directors of the Continuing Company, until such time as their successors have been elected and qualified, shall consist of the current Board of Directors of F&M. The officers of F&M immediately prior to the Effective Date shall continue as the officers of the Continuing Company.
- 2.3 <u>Capital Structure</u>. The amount of capital stock of the Continuing Company shall not be less than the capital stock of F&M immediately prior to the Effective Date increased by the amount of capital stock issued in accordance with Section 3 hereof.
- 2.4 <u>Articles of Incorporation and Code of Regulations</u>. The Articles of Incorporation and the Code of Regulations of the Continuing Company shall be those of F&M immediately prior to the Effective Date until the same shall be further amended as provided therein or by law.

- 2.5 <u>Assets and Liabilities</u>. The title to all assets, real estate and other property owned by F&M and LBI shall vest in the Continuing Company without reversion or impairment. All liabilities of LBI shall be assumed by the Continuing Company.
- 2.6 Additional Actions. If, at any time after the Effective Date, the Continuing Company shall consider or be advised that any further deeds, assignments or assurances in law or any other acts are necessary or desirable (a) to vest, perfect or confirm, of record or otherwise, in the Continuing Company its right, title or interest in, to or under any of the rights, properties or assets of LBI or the Bank, or (b) otherwise carry out the purposes of this Agreement, LBI and the Bank and their respective officers and directors shall be deemed to have granted to the Continuing Company an irrevocable power of attorney to execute and deliver all such deeds, assignments or assurances in law and to do all acts necessary or proper to vest, perfect or confirm title to and possession of such rights, properties or assets in the Continuing Company and otherwise to carry out the purposes of this Agreement, and the officers and directors of the Continuing Company are authorized in the name of LBI or the Bank or otherwise to take any and all such action.

SECTION 3

CONSIDERATION TO BE DISTRIBUTED

- 3.1 <u>Consideration</u>. Upon and by reason of the Merger becoming effective, the shareholders of LBI of record on the Effective Date shall be entitled to receive in exchange for each LBI common share, \$10.00 par value, ("LBI Common Stock") held: (i) 1,830.00 (the "Exchange Ratio") shares of F&M common stock ("F&M Common Stock"), and (ii) \$8,465.00 in cash which cash amount is subject to a potential reduction as set forth in Section 5.10 of the Disclosure Letter. The F&M Common Stock and cash to be exchanged for each share of LBI Common Stock is collectively referred to herein as the "Merger Consideration". The Exchange Ratio shall be subject to adjustment as set forth in Section 3.3, but shall otherwise remain fixed and no adjustment to the Merger Consideration shall be made as a result of changes in the stock price of the F&M Common Stock.
- 3.2 <u>Fractional F&M Common Shares</u>. Certificates for fractional shares of F&M Common Stock shall not be issued in respect of fractional interests arising from the Exchange Ratio. Each holder of LBI Common Stock who would have otherwise been entitled to a fraction of a share of F&M Common Stock, upon surrender of all such shareholder's certificates representing LBI Common Stock, shall be paid in cash (without interest), an amount rounded to the nearest whole cent, determined by multiplying the F&M Average Price (as defined below) by the fractional share of F&M Common Stock to which such holder of LBI Common Stock would otherwise be entitled. No such holder of LBI Common Stock shall be entitled to dividends, voting rights, or any other rights in respect of any fractional share. The term "F&M Average Price" shall mean the average closing price of a share of F&M Common Stock for the ten (10) days that F&M Common Stock trades on the NASDAQ Capital Market preceding the fourth (4th) calendar day prior to the Effective Date. The F&M Average Price shall be appropriately and proportionately adjusted to reflect any share adjustment as contemplated by Section 3.3 hereof.
- 3.3 <u>Recapitalization</u>. If, between the date of this Agreement and the Effective Date, F&M declares a stock dividend, combination, subdivision, or split with respect to the outstanding F&M Common Stock, or takes any similar recapitalization action, and the record date therefor is prior to the Effective Date, then the Exchange Ratio shall be adjusted so that each holder of LBI Common Stock shall receive such number of shares of F&M Common Stock as represents the same percentage of outstanding shares of F&M Common Stock at the Effective Date as would have been represented by the number of shares of F&M Common Stock such shareholder would have received if the recapitalization had not occurred.

3.4 Distribution of F&M's Common Stock.

(a) Each share of F&M Common Stock outstanding immediately prior to the Effective Date shall remain outstanding unaffected by the Merger. Any shares of LBI Common Stock held by F&M shall be cancelled by virtue of the Merger and not be exchanged for shares of F&M Common Stock.

On or prior to the Effective Date, F&M shall (i) authorize the issuance of and shall make available to Computershare Trust Company or such other exchange agent selected by F&M (the "Exchange Agent"), for the benefit of the registered shareholders of LBI Common Stock for exchange in accordance with this Section 3, certificates for shares (or book entry of shares) of F&M Common Stock (the "F&M Stock Certificates") to be issued pursuant to Section 3.1, and (ii) shall deposit with the Exchange Agent sufficient cash for payment of the cash portion of the Merger Consideration and cash in lieu of any fractional shares of F&M Common Stock in accordance with Section 3.2. Such F&M Stock Certificates and cash are referred to in this Section 3 as the "Exchange Fund." F&M shall be solely responsible for the payment of any fees and expenses of the Exchange Agent.

(b) Within three (3) business days following the Effective Date, the Exchange Agent shall mail to each holder of LBI Common Stock a letter of transmittal (the "Letter of Transmittal") providing (i) that delivery shall be effected and risk of loss of title to the certificates representing LBI Common Stock shall pass only upon delivery of the certificates to the Exchange Agent and (ii) instructions as to the transmittal to the Exchange Agent of certificates representing shares of LBI Common Stock and the issuance of shares of F&M Common Stock and cash in exchange therefor pursuant to the terms of this Agreement. Distribution of shares of F&M Common Stock Certificates, the cash portion of the Merger Consideration, and cash payments in lieu of fractional shares shall be made by the Exchange Agent to each former holder of LBI Common Stock within ten (10) business days following the later of the Effective Date or the date of such shareholder's delivery to the Exchange Agent of such shareholder's certificates representing LBI Common Stock, accompanied by a properly completed and executed Letter of Transmittal. Interest shall not accrue or be payable with respect to any cash payments.

Following the Effective Date, stock certificates representing LBI Common Stock shall be converted to, and deemed to evidence only the right to receive such number of shares of F&M Common Stock and cash as determined in accordance with Sections 3.1, 3.2 and 3.3 above (for all corporate purposes other than the payment of dividends) and cash for fractional shares, as applicable. No dividends or other distributions otherwise payable subsequent to the Effective Date on shares of F&M Common Stock shall be paid to any shareholder entitled to receive the same until such shareholder has surrendered such shareholder's certificates for LBI Common Stock to the Exchange Agent in exchange for the Merger Consideration. Upon surrender or compliance with the provisions of Section 3.4(c), there shall be paid to the record holder of F&M Common Stock the amount of all dividends and other distributions, without interest thereon, withheld with respect to such common stock.

- (c) From and after the Effective Date, there shall be no transfers on the stock transfer books of LBI of any shares of LBI Common Stock.
- (d) Any portion of the Exchange Fund that remains unclaimed by the holders of LBI Common Stock for twelve (12) months after the Effective Date shall be paid, distributed, or otherwise released to F&M, or its successors in interest. Any shareholders of LBI who have not theretofore complied with this Section 3 shall thereafter look only to F&M, or its successors in interest, for the issuance of shares of F&M Common Stock, payment of cash and any unpaid dividends and distributions on F&M Common Stock deliverable in respect of each share of LBI Common Stock such shareholder holds as determined pursuant to this Agreement. Notwithstanding the foregoing, none of F&M, the Exchange Agent or any other person shall be liable to any former holder of shares of LBI Common Stock for any amount delivered in good faith to a public official pursuant to applicable abandoned property, escheat or similar laws.
- (e) F&M shall be entitled to rely upon the stock transfer books of LBI to establish the persons entitled to receive the Merger Consideration, which books, in the absence of actual knowledge by F&M of any adverse claim thereto, shall be conclusive with respect to the ownership of such stock.
- (f) With respect to any certificate for LBI Common Stock which has been lost, stolen, or destroyed, the Exchange Agent and F&M shall be authorized to issue F&M Common Stock and cash to the registered owner of such certificate upon receipt of an affidavit of lost stock certificate, in form and substance reasonably satisfactory to the Exchange Agent and F&M, and, if required by the Exchange Agent or F&M in their sole discretion, upon the posting of a bond in such amount as the Exchange Agent or F&M may determine is reasonably necessary as indemnity against any claim that may be made in regard to any lost, stolen or destroyed certificate representing LBI Common Stock, with any costs incurred at the shareholder's expense.

SECTION 4

DISSENTING SHAREHOLDERS

Notwithstanding anything in this Agreement to the contrary, any issued and outstanding shares of LBI Common Stock held by a person (a "Dissenting Shareholder") who has not voted in favor of, or consented to, the adoption of this Agreement and has complied with all the provisions of the Indiana Business Corporation Law concerning the rights of holders of shares of LBI Common Stock to require payment in cash of the "fair value" of such LBI Common Stock (the "Dissenting Shares"), in accordance with Indiana Code § 23-1-44, as amended, shall not be converted into the right to receive the Merger Consideration and cash for fractional shares as determined in accordance with Sections 3.1 and 3.2 above, but shall become the right to receive such consideration as may be determined to be due to such Dissenting Shareholder pursuant to the procedures set forth in Indiana Code § 23-1-44, as amended. If such Dissenting Shareholder

withdraws its demand for "fair value" or fails to perfect or otherwise loses its rights as a dissenting shareholder, in any case under the Indiana Business Corporation Law, each of such Dissenting Shareholder's shares of LBI Common Stock shall thereupon be treated as though such shares of LBI Common Stock had been converted into the right to receive the Merger Consideration pursuant to Sections 3.1 and 3.2. LBI shall notify F&M of each shareholder who asserts rights as a Dissenting Shareholder following receipt of such Dissenting Shareholder's written demand delivered as provided in Indiana Code § 23-1-44, as amended. Prior to the Effective Date, LBI shall not, except with the prior written consent of F&M, voluntarily make any payment, settle or commit to settle, or offer to settle any rights of a Dissenting Shareholder asserted under Indiana Code § 23-1-44, as amended.

SECTION 5

REPRESENTATIONS AND WARRANTIES OF LBI

LBI hereby makes the representations and warranties set forth below to F&M with respect to itself and the Bank. For the purposes of this Agreement, "LBI Disclosure Letter" is defined as the letter referencing Section 5 of this Agreement which shall be prepared by LBI and delivered to F&M contemporaneously with the execution of this Agreement.

5.1 <u>Organization and Authority</u>. LBI and the Bank are each a corporation duly organized and validly existing under the laws of the State of Indiana. Each of LBI and the Bank have the power and authority (corporate and otherwise) to conduct their respective businesses in the manner and by the means utilized as of the date hereof. LBI's only subsidiary is the Bank. The Bank has no subsidiaries. The Bank is subject to primary federal regulatory supervision and regulation by the Board of Governors of the Federal Reserve System (the "**Federal Reserve Board**").

5.2 Authorization.

(a) LBI has the corporate power and authority to enter into this Agreement and to carry out its obligations hereunder, subject to satisfaction of the conditions precedent in Section 9. This Agreement, when executed and delivered by all parties, will have been duly authorized and will constitute a valid and binding obligation of LBI, subject to the conditions precedent set forth in Section 9 hereof, enforceable in accordance with its terms except to the extent limited by insolvency, reorganization, liquidation, readjustment of debt or other laws of general application relating to or affecting the enforcement of creditors' rights. The respective Boards of Directors of LBI and the Bank, and LBI as the sole shareholder of the Bank, have approved the Merger and the Bank Merger pursuant to the terms and conditions of this Agreement and the Bank Merger Agreement. The Board of Directors of LBI has agreed to recommend to LBI's shareholders that they adopt this Agreement and approve the transactions described herein subject to Section 7.5 hereof.

(b) Except as set forth in the LBI Disclosure Letter, neither the execution of this Agreement, nor the consummation of the transactions contemplated hereby, subject to the conditions precedent set forth in Section 9 hereof, does or will (i) conflict with, result in a breach of, or constitute a default under LBI's or the Bank's organizational documents; (ii) conflict with, result in a breach of, or constitute a default under any federal, foreign, state or local law, statute, ordinance, rule, regulation or court or administrative order or decree, or any note, bond, indenture, loan, mortgage, security agreement, contract, arrangement or commitment, to which LBI or the Bank is subject or bound, the result of which would have a Material Adverse Effect; (iii) result in the creation of, or give any person, corporation or entity the right to create, any lien, charge, encumbrance, security interest, or any other rights of others or other adverse interest upon any right, property or asset of LBI or the Bank; (iv) terminate, or give any person, corporation or entity the right to terminate, amend, abandon, or refuse to perform, any note, bond, indenture, loan, mortgage, security agreement, contract, arrangement or commitment to which LBI or the Bank is subject or bound, the result of which would have a Material Adverse Effect; or (v) accelerate or modify, or give any party thereto the right to accelerate or modify, the time within which, or the terms according to which, LBI or the Bank is to perform any duties or obligations or receive any rights or benefits under any note, bond, indenture, loan, mortgage, security agreement, contract, arrangement or commitment.

For the purpose of this Agreement, a "Material Adverse Effect" means any effect, circumstance, occurrence or change that (i) is material and adverse to the financial position, results of operations or business of LBI and the Bank taken as a whole, or F&M and F&M Bank taken as a whole, as applicable, or (ii) would materially impair the ability of LBI or F&M, as applicable, to perform its obligations under this Agreement; provided, however, that a Material Adverse Effect shall not be deemed to include the impact of (a) changes in banking and similar laws of general applicability to banks or their holding companies or interpretations thereof by courts or governmental authorities, (b) changes in generally accepted accounting principles ("GAAP") or regulatory accounting requirements applicable to banks or their holding companies generally, (c) any modifications or changes to valuation policies and practices in connection with the Merger or restructuring charges taken in connection with the Merger, in each case in accordance with GAAP, (d) effects of any action taken with the prior written consent of the other party hereto, (e) changes in the general level of interest rates (including the impact on the securities portfolios of LBI and the Bank, or F&M and F&M Bank, as applicable) or conditions or circumstances relating to or that affect either the United States economy, financial or securities markets or the banking industry, generally, (f) changes resulting from expenses (such as legal, accounting and investment bankers' fees) incurred in connection with this Agreement or the transactions contemplated herein, including without limitation payment of any amounts due to, or the provision of any benefits to, any officers or employees under agreements, plans or other arrangements in existence on the date of or contemplated by this Agreement and disclosed to F&M, (g) the impact of the announcement of this Agreement and the transactions contemplated hereby, and compliance with this Agreement on the business, financial condition or results of operations of LBI and the Bank, or F&M and F&M Bank, as applicable, and (h) the occurrence of any military or terrorist attack within the United States or any of its possessions or offices; provided that in no event shall a change in the trading price of the F&M Common Stock, by itself, be considered to constitute a Material Adverse Effect on F&M (it being understood that the foregoing proviso shall not prevent or otherwise affect a determination that any effect underlying such decline has resulted in a Material Adverse Effect).

- (c) Other than the filing of a Certificate of Merger with the Ohio Secretary of State and Articles of Merger with the Indiana Secretary of State for the Merger and the Bank Merger and in connection or in compliance with the banking regulatory approvals contemplated by Section 9.4 and federal and state securities laws and the rules and regulations promulgated thereunder, no notice to, filing with, authorization of, exemption by, or consent or approval of, any public body or authority is necessary for the consummation by LBI of the transactions contemplated by this Agreement.
- (d) Other than those filings, authorizations, consents and approvals referenced in Section 5.2(c) above and except as set forth in the LBI Disclosure Letter, no notice to, filing with, authorization of, exemption by, or consent or approval of, any third party is necessary for the consummation by LBI or the Bank of the transactions contemplated by this Agreement, except for such authorizations, exemptions, consents or approvals, the failure of which to obtain, would not be reasonably likely to result in a Material Adverse Effect.

5.3 Capitalization.

- (a) As of the date of this Agreement, LBI has 1,000,000 shares, \$10.00 par value per share, of LBI Common Stock authorized, 1,000 shares of which are issued and outstanding. Such issued and outstanding shares of LBI Common Stock have been duly and validly authorized by all necessary corporate action of LBI, are validly issued, fully paid and nonassessable and have not been issued in violation of any preemptive rights of any shareholders. LBI has no capital stock authorized, issued or outstanding other than as described in this Section 5.3(a) and, except as set forth in the LBI Disclosure Letter, LBI has no intention or obligation to authorize or issue additional shares of its capital stock.
- (b) As of the date of this Agreement, the Bank has 275,000 shares of common stock, \$10.00 par value per share, authorized and outstanding, all of which are held beneficially and of record by LBI. Such issued and outstanding shares of Bank common stock have been duly and validly authorized by all necessary corporate action of the Bank, are validly issued, fully paid and nonassessable, and have not been issued in violation of any preemptive rights of any Bank shareholder. All of the issued and outstanding shares of Bank common stock are owned by LBI free and clear of all liens, pledges, charges, claims, encumbrances, restrictions, security interests, options and preemptive rights and of all other rights of any other person, corporation or entity with respect thereto. The Bank has no capital stock authorized, issued or outstanding other than as described in this Section 5.3(b) and has no intention or obligation to authorize or issue any other shares of capital stock.
- (c) Except as set forth on the LBI Disclosure Letter there are no options, commitments, calls, agreements, understandings, arrangements or subscription rights regarding the issuance, purchase or acquisition of capital stock, or any securities convertible into or representing the right to purchase or otherwise receive the capital stock, equity interests, or any debt securities, of LBI or the Bank by which LBI or the Bank is or may become bound. Except as set forth on the LBI Disclosure Letter, neither LBI nor the Bank has any outstanding contractual or other obligation to repurchase, redeem or otherwise acquire any of its respective outstanding shares of capital stock or equity interests, as applicable.

- (d) Except as set forth in the LBI Disclosure Letter, to the knowledge of LBI's Management (as defined below), no person or entity beneficially owns five percent (5%) or more of the total issued and outstanding LBI Common Stock.
- 5.4 <u>Organizational Documents</u>. Except as set forth on the LBI Disclosure Letter, the respective Articles of Incorporation and By-Laws of LBI and the Bank have been delivered to F&M and represent true, accurate and complete copies of such corporate (or organizational) documents of LBI and the Bank in effect as of the date of this Agreement.
- 5.5 Compliance with Law. To the knowledge of LBI's Management (as defined below) neither LBI nor the Bank has engaged in any activity nor taken or omitted to take any action which has resulted or could reasonably be expected to result, in the violation of any local, state, federal or foreign law, statute, rule, regulation or ordinance or of any order, injunction, judgment or decree of any court or government agency or body, the violation of which could reasonably be expected to have a Material Adverse Effect on LBI. LBI and the Bank possess all licenses, franchises, permits and other authorizations necessary for the continued conduct of their respective businesses without material interference or interruption, except where the failure to possess such licenses or other authorizations would not be reasonably expected to have a Material Adverse Effect on LBI, and such licenses, franchises, permits and authorizations shall be transferred to F&M on the Effective Date without any material restrictions or limitations thereon or the need to obtain any consents of third parties, except as otherwise set forth in the LBI Disclosure Letter. Neither LBI nor the Bank is subject to any agreement, commitment or understanding with, or order and directive of, any regulatory agency or government authority with respect to the business or operations of LBI or the Bank. The Bank has not received any notice of enforcement actions since January 1, 2015 from any regulatory agency or government authority relating to its compliance with the Bank Secrecy Act, the Truth-in-Lending Act, the Community Reinvestment Act, the Gramm-Leach-Bliley Act of 1999, the USA Patriot Act, the International Money Laundering Abatement and Financial Anti-Terrorism Act of 2001, the Sarbanes-Oxley Act of 2002, the Dodd-Frank Wall Street Reform and Consumer Protection Act or any laws with respect to the protection of the environment or the rules and regulations promulgated thereunder. LBI has not received any notice of enforcement actions since January 1, 2015, from any regulatory agency or government authority relating to its compliance with any securities laws applicable to LBI. The Bank received a rating of "satisfactory" or better in its most recent examination or interim review with respect to the Community Reinvestment Act.
- 5.6 Accuracy of Statements. No information which has been or shall be supplied by LBI with respect to its businesses, operations and financial condition for inclusion in the proxy statement, registration statement, and regulatory applications relating to the Merger or the Bank Merger contains or shall contain (in the case of information relating to the proxy statement at the time it is mailed and for the regulatory applications and registration statement, and each amendment or supplement thereto, if any, at the time it becomes effective) any untrue statement of a material fact or omits or shall omit to state a material fact necessary to make the statements contained therein not misleading.

5.7 <u>Litigation and Pending Proceedings</u>. Except as set forth in the LBI Disclosure Letter, there are no claims of any kind, nor any action, suits, proceedings, arbitrations or investigations pending or, to the knowledge of LBI's Management, threatened in any court or before any government agency or body, arbitration panel or otherwise (nor does LBI's Management have any knowledge of a basis for any claim, action, suit, proceeding, arbitration or investigation) which could reasonably be expected to have a Material Adverse Effect. To the knowledge of LBI's Management, there are no uncured violations, criticisms or exceptions, or violations with respect to which refunds or restitutions may be required, cited in any report, correspondence or other communication to LBI or the Bank as a result of an examination by any regulatory agency or body which could reasonably be expected to have a Material Adverse Effect.

5.8 Financial Statements.

- (a) LBI's consolidated audited balance sheets as of the end of the two (2) fiscal years ended December 31, 2017 and 2016, the unaudited consolidated balance sheet for the six months ended June 30, 2018 and the related consolidated statements of income, shareholders' equity and cash flows for the years or period then ended (hereinafter collectively referred to as the "Financial Information"), present fairly the consolidated financial condition or position of LBI as of the respective dates thereof and the consolidated results of operations of LBI for the respective periods covered thereby and have been prepared in conformity with GAAP applied on a consistent basis.
- (b) All loans reflected in the Financial Information and which have been made, extended or acquired since December 31, 2017 (i) have been made for good, valuable and adequate consideration in the ordinary course of business; (ii) constitute the legal, valid and binding obligation of the obligor and any guarantor named therein; (iii) are evidenced by notes, instruments or other evidences of indebtedness which are true, genuine and what they purport to be; and (iv) to the extent that the Bank has a security interest in collateral or a mortgage securing such loans, are secured by perfected security interests or mortgages naming the Bank as the secured party or mortgagee, except for such unperfected security interests or mortgages naming the Bank as secured party or mortgagee which, on an individual loan basis, would not materially adversely affect the value of any such loan and the recovery of payment on any such loan if the Bank is not able to enforce any such security interest or mortgage.
- 5.9 Absence of Certain Changes. Except for events and conditions relating to the business and interest rate environment in general, the accrual or payment of Merger-related expenses, or as set forth in the LBI Disclosure Letter, since December 31, 2017, no events have occurred which could reasonably be expected to have a Material Adverse Effect. Except as set forth in the LBI Disclosure Letter, between the period from December 31, 2017 to the date of this Agreement, LBI and the Bank have carried on their respective businesses in the ordinary and usual course consistent with their past practices (excluding the incurrence of fees and expenses of professional advisors related to this Agreement and the transactions contemplated hereby) and there has not been any declaration, setting aside or payment of any dividend or other distribution (whether in cash, stock or property) with respect to LBI's Common Stock (other than normal quarterly cash dividends or distributions consistent with past practice) or any split, combination or reclassification of LBI Common Stock or, with the exception of the issuance of shares in connection with the exercise of stock options or the vesting of any performance stock units, any issuance or the authorization of any issuance of any securities in respect of, or in lieu of, or in substitution for LBI Common Stock.

5.10 <u>Absence of Undisclosed Liabilities</u>. Except as set forth in the LBI Disclosure Letter, neither LBI nor the Bank has any liabilities, whether accrued, absolute, contingent, or otherwise, existing or arising out of any transaction or state of facts existing on or prior to the date hereof, except (a) as and to the extent disclosed, reflected or reserved against in the Financial Information, (b) any agreement, contract, obligation, commitment, arrangement, liability, lease or license which individually is less than Twenty-Five Thousand and 00/100 Dollars (\$25,000.00) per year and which may be terminated within one year from the date of this Agreement, (c) liabilities incurred since December 31, 2017 in the ordinary course of business consistent with past practice that either alone or when considered with all similar liabilities, have not had or would not reasonably be expected have a Material Adverse Effect on LBI, (d) liabilities incurred for reasonable legal, accounting, financial advising fees and out-of-pocket expenses or fees in connection with the transactions contemplated by this Agreement, and (e) unfunded loan commitments and standby letters of credit made in the ordinary course of the Bank's business consistent with past practices.

5.11 <u>Title to Assets</u>.

- (a) Except as set forth on the LBI Disclosure Letter, LBI and the Bank have good and marketable title to all personal property reflected in the December 31, 2017 Financial Information, good and marketable title to all other properties and assets which LBI or the Bank purports to own, good and marketable title to or right to use by terms of any lease or contract all other property used in LBI's or the Bank's business, and good and marketable title to all property and assets acquired since December 31, 2017, free and clear of all mortgages, liens, pledges, restrictions, security interests, charges, claims or encumbrances of any nature, except such minor imperfections of title, if any, as do not materially detract from the value of or interfere with the use of the property and which would not have a Material Adverse Effect.
- (b) The operation by LBI or the Bank of such properties and assets is in material compliance with all applicable laws, ordinances, rules and regulations of any governmental authority or third party having jurisdiction over such use except for such noncompliance that would not have a Material Adverse Effect.

5.12 Loans and Investments.

(a) Except as set forth in the LBI Disclosure Letter, there is no loan of the Bank in excess of One Hundred Thousand and 00/100 Dollars (\$100,000.00) that, as of June 30, 2018, with respect to classified loans, and July 11, 2018, with respect to special mention loans and marginally pass loans, that (i) has been classified by LBI, applying applicable regulatory examination standards, as "Other Loans Specially Mentioned," "Substandard," "Doubtful" or "Loss;" (ii) has been identified by accountants or auditors (internal or external) as having a significant risk of uncollectibility, or (iii) has been identified by LBI Management to be ninety (90) days or more past due with respect to principal or interest or has placed on nonaccrual status.

- (b) The reserves for loan and lease losses and the carrying value for other real estate owned which are shown on each of the balance sheets contained in the Financial Information were adequate in the judgment of LBI's Management and consistent with applicable bank regulatory standards and under GAAP to provide for losses, net of recoveries relating to loans and leases previously charged off, on loans and leases outstanding and other real estate owned (including accrued interest receivable) as of the applicable date of such balance sheet.
- (c) Except as set forth in the LBI Disclosure Letter, none of the investments reflected in the Financial Information and none of the investments made by LBI or the Bank since December 31, 2017 are subject to any restrictions, whether contractual or statutory, which materially impairs the ability of LBI or the Bank to dispose freely of such investment at any time. Except as set forth in the LBI Disclosure Letter, LBI or the Bank is not a party to any repurchase agreements with respect to securities.

5.13 Employee Benefit Plans.

- (a) The LBI Disclosure Letter contains a list identifying each "employee benefit plan," as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), which (i) is subject to any provision of ERISA, and (ii) is currently maintained, administered or contributed to by LBI, the Bank or any other entity, trade or business that, together with LBI, would be treated as a single employer under the provisions of Sections 414(b), (c), (m) or (o) of the Code ("LBI ERISA Affiliate"), and covers any employee, director or former employee or director of LBI, the Bank or any LBI ERISA Affiliate under which LBI or any LBI ERISA Affiliate has any liability. The LBI Disclosure Letter also contains a list of all "employee benefit plans," as defined under ERISA which have been terminated by LBI, the Bank or any LBI ERISA Affiliate since January 1, 2013. Copies of such plans (and, if applicable, related trust agreements or insurance contracts) and all amendments thereto and written interpretations thereof have been furnished to F&M together with the three (3) most recent annual reports (Form 5500) prepared in connection with any such plan and the current summary plan descriptions (and any summary of material modifications thereto). Such plans are hereinafter referred to individually as an "Employee Plan" and collectively as the "Employee Plans." The Employee Plans which individually or collectively would constitute an "employee pension benefit plan" as defined in Section 3(2)(A) of ERISA are identified as such in the list referred to above.
- (b) The Employee Plans have been operated in material compliance with all applicable laws, regulations, rulings and other requirements, as well as pursuant to the terms of their governing documents (to the extent consistent with ERISA).
- (c) Except as disclosed in the LBI Disclosure Letter, no "prohibited transaction," as defined in Section 406 of ERISA or Section 4975 of the Code, for which no statutory or administrative exemption exists, and no "reportable event," as defined in Section 4043(c) of ERISA, for which a notice is required to be filed, has occurred with respect to any Employee Plan that could subject LBI to material taxes or penalties.

(d) Neither LBI, the Bank nor any LBI ERISA Affiliate has any material liability to the Pension Benefit Guaranty Corporation ("**PBGC**"), to the Internal Revenue Service ("**IRS**"), to the Department of Labor ("**DOL**"), to the Employee Benefits Security Administration, with respect to any Employee Plan, except for routine premium payments to the PBGC.

No purchase of shares of LBI Common Stock or other securities issued by LBI by any Employee Plan holding shares of LBI Common Stock or other securities of LBI, including but not limited to the Bank of Geneva Employee Stock Ownership Plan (the "ESOP"), has adversely affected the tax qualification of such Employee Plan or failed to satisfy all of the requirements for the prohibited transaction exemption provided by Section 408(e) of ERISA. Each loan to the ESOP or such other Employee Plan acquiring shares of LBI Common Stock or other employer securities which has been made by or guaranteed by LBI, the Bank or any other disqualified person in connection with any purchase of such shares by the Employee Plan (an "ESOP Loan"), satisfied each of the requirements of the prohibited transaction exemption provided in Section 408(b)(3) of ERISA, Section 4975(d)(3) and Treasury Regulation Section 54-4975-7(b), and, in particular, all shares of LBI Common Stock purchased by the ESOP or any other Employee Plan were purchased for no more than "adequate consideration" within the meaning Section 3(18) of ERISA, as determined on the basis of a stock valuation prepared by an "independent appraiser" (as this term is defined in Section 401(a)(28)(C) of the Code) satisfying all requirements of Sections 3(18) and 408(e) of ERISA and applicable DOL regulations.

The ESOP has at all times been maintained and administered in accordance with and met the requirements of Section 409(p) of the Code, no plan year of the ESOP has been a "non-allocation year" (as this term is defined in Section 409(p) of the Code), and LBI and the Bank have performed adequate testing and maintained adequate records to demonstrate compliance with the requirements of Section 409(p) of the Code and the Treasury Regulations promulgated thereunder.

- (e) Except as disclosed in the LBI Disclosure Letter, no "fiduciary," as defined in Section 3(21) of ERISA, of an Employee Plan has failed to comply with the requirements of Section 404 of ERISA in such a way as to cause material liability to LBI, the Bank or any LBI ERISA Affiliate.
- (f) Each of the Employee Plans which is intended to be qualified under Section 401(a) of the Code has been timely amended to comply in all material respects with the applicable requirements of the Code, including, in the case of the ESOP, Sections 409 and 4975 of the Code. LBI or the Bank has sought and received a favorable determination letter from the IRS ruling that the ESOP qualifies as an employee stock ownership plan described in Section 4975(e)(7) of the Code. Except as set forth in the LBI Disclosure Letter, LBI and/or any LBI ERISA Affiliate, as applicable, sought and received favorable determination letters from the IRS (or are otherwise relying on an opinion letter issued to a prototype plan sponsor) and has furnished to F&M copies of the most recent IRS determination letters with respect to any such Employee Plan that is intended to be qualified under Section 401(a) of the Code, and, in the case of the ESOP, Section 4975(e)(7) of the Code.

- (g) Except as disclosed in the LBI Disclosure Letter, no Employee Plan maintained or contributed to by LBI, the Bank or any ERISA Affiliate is a defined benefit pension plan subject to Title IV of ERISA or the minimum funding standards under Sections 412 and 430 of the Code and Section 302 of ERISA. No facts or circumstances exist that may subject LBI, the Bank, or any LBI ERISA Affiliate, to any liability under Sections 4062, 4063 or 4064 of ERISA. Except as disclosed in the LBI Disclosure Letter, there exist no facts or circumstances which could subject LBI, or any LBI ERISA Affiliate thereof, to withdrawal liability within the meaning of Section 4201 of ERISA or to contingent withdrawal liability under Section 4204 of ERISA. Neither LBI nor any LBI ERISA Affiliate ever has been a party to a transaction within the meaning of Section 4212(c) of ERISA.
- (h) No claims involving an Employee Plan (other than normal benefit claims) have been filed in a court of law or, to the knowledge of LBI's Management, have been threatened to be filed in a court of law.
- (i) For all calendar months since January 1, 2015 in which the Bank, LBI or any ERISA Affiliate was an "applicable large employer" subject to Section 4980H of the Code, LBI, the Bank or such ERISA Affiliate has offered to substantially all of its full-time employees (and their dependents) the opportunity to enroll in minimum essential coverage under an eligible employer-sponsored group plan which complies in all material respects with the requirements of the Affordable Care Act. To the knowledge of LBI's Management, no event has occurred that would cause the imposition of the tax described in Section 4980B of the Code on LBI. To the knowledge of LBI's Management, LBI has materially complied with all requirements of Section 601 of ERISA, as applicable, with respect to any Employee Plan. Except as set forth in the LBI Disclosure Letter or as required by applicable law, neither LBI nor any LBI ERISA Affiliate has any present or future liability in respect of post-retirement health and medical benefits for former employees or directors of LBI, the Bank or any LBI ERISA Affiliate.
- (j) The LBI Disclosure Letter contains a list of each employment, severance or other similar contract, arrangement or policy and each plan or arrangement (written or oral) providing for insurance coverage (including any self-insured arrangements), workers' compensation, disability benefits, supplemental unemployment benefits, vacation benefits, retirement benefits or deferred compensation, profit sharing, bonuses, stock options, stock appreciation or other forms of incentive compensation or post-retirement insurance, compensation or benefits which (i) is not an Employee Plan, (ii) is established, maintained or contributed to, as the case may be, by LBI or the Bank and (iii) covers any employee, director or former employee or director of LBI or the Bank. Such contracts, plans and arrangements as are described above, copies or descriptions of all of which have been furnished previously to F&M, are hereinafter referred to collectively as the "Benefit Arrangements." Each of the Benefit Arrangements has been maintained in compliance in all material respects with its terms and with the requirements prescribed by any and all statutes, orders, rules and regulations which are applicable to such Benefit Arrangements.

- (k) Except as set forth in the LBI Disclosure Letter, there has been no amendment to, written interpretation or announcement (whether or not written) by LBI, the Bank or any LBI ERISA Affiliate relating to, or change in employee participation or coverage under, any Employee Plan or Benefit Arrangement administered by LBI or any LBI ERISA Affiliate which would increase materially the expense of maintaining such Employee Plans or Benefit Arrangements above the level of the expense incurred in respect thereof for the fiscal year ended December 31, 2017.
- (l) Except as otherwise provided in the LBI Disclosure Letter, the transactions contemplated by the Agreement will not cause acceleration of vesting in, or payment of, any material benefits under any Employee Plan or Benefit Arrangement and will not otherwise materially accelerate or increase any obligation under any Employee Plan or Benefit Arrangement.
- (m) With respect to any Employee Plan which is a deferred compensation plan that is subject to Section 409A of the Code, such plan has been identified on the LBI Disclosure Letter and, except as otherwise set forth in the LBI Disclosure Letter, has been operated in accordance with, and is in documentary compliance with Section 409A of the Code and the guidance issued thereunder.

5.14 Obligations to Employees.

LBI is in compliance in all material respects with all applicable laws respecting employment including discrimination in employment, terms and conditions of employment, worker classification (including the proper classification of workers as independent contractors and consultants and of employees as exempt or non-exempt, in each case, under the Fair Labor Standards Act of 1938, as amended (the "Fair Labor Standards Act"), and any similar applicable law), wages, hours and occupational safety and health and employment practices, including the Immigration Reform and Control Act, and is not engaged in any unfair labor practice.

Neither the Bank nor LBI is or was a party to, bound by or subject to, or is currently negotiating, any collective bargaining agreement or understanding with a labor union or organization. Except as set forth in the LBI Disclosure Letter, all employees of the Bank and LBI are employed on an "at will" basis, and neither LBI nor the Bank has any contractual or legal obligations that would prevent the Bank or LBI from termination the employment of any employee of the Bank or LBI without liability for severance pay.

Except as set forth in the LBI Disclosure Letter, all accrued obligations and liabilities of LBI and the Bank, whether arising by operation of law, by contract or by past custom, for payments to trust or other funds, to any government agency or body or to any individual director, officer, employee or agent (or his heirs, legatees or legal representative) with respect to unemployment compensation or social security benefits and all pension, retirement, savings, stock purchase, stock bonus, stock ownership, stock option, stock appreciation rights or profit sharing plan, any employment, deferred compensation, consultant, bonus or collective bargaining agreement or group insurance contract or other incentive, welfare or employee benefit plan or agreement maintained by LBI or the Bank for their current or former directors, officers, employees and agents have been and are being paid to the extent required by law or by the plan or contract, and adequate actuarial accruals and/or reserves for such payments have been and are being made by LBI or the Bank in accordance with generally accepted accounting and actuarial principles, except

where the failure to pay any such accrued obligations or liabilities or to maintain adequate accruals and/or reserves for payment thereof would not have a Material Adverse Effect. Except as set forth in the LBI Disclosure Letter, all obligations and liabilities of LBI and the Bank, whether arising by operation of law, by contract, or by past custom, for all forms of compensation which are or may be payable to their current or former directors, officers, employees or agents have been and are being paid, and adequate accruals and/or reserves for payment therefore have been and are being made in accordance with GAAP, except where the failure to pay any such obligations and liabilities or to maintain adequate accruals and/or reserves for payment thereof would not have a Material Adverse Effect. All accruals and reserves referred to in this Section 5.14 are correctly and accurately reflected and accounted for in the books, statements and records of LBI and the Bank, except where the failure to correctly and accurately reflect and account for such accruals and reserves would not have a Material Adverse Effect.

- 5.15 Taxes, Returns and Reports. Except as set forth in the LBI Disclosure Letter, LBI and the Bank have (a) duly filed all federal, state, local and foreign tax returns of every type and kind required to be filed by them as of the date hereof, and each return is true, complete and accurate in all material respects; (b) paid all material taxes, assessments and other governmental charges due and payable or claimed to be due and payable upon them or any of their income, properties or assets; and (c) not requested an extension of time for any such payments (which extension is still in force). Except for taxes not yet due and payable, the reserve for taxes on the Financial Information is adequate to cover all of LBI's and the Bank's tax liabilities (including, without limitation, income taxes and franchise fees) that may become payable in future years with respect to any transactions consummated prior to December 31, 2017. Neither LBI nor the Bank has or will have, any liability for taxes of any nature for or with respect to the operation of their business from December 31, 2017, up to and including the Effective Date, except to the extent reflected on their Financial Information or on financial statements of LBI or the Bank subsequent to such date and as set forth in the LBI Disclosure Letter. Neither LBI nor the Bank has received written notice that it is currently under audit by any state or federal taxing authority. Except as set forth in the LBI Disclosure Letter, none of the federal, state, or local tax returns of LBI or the Bank have been audited by any taxing authority during the past five (5) years.
- 5.16 <u>Deposit Insurance</u>. The deposits of the Bank are insured by the Federal Deposit Insurance Corporation ("**FDIC**") in accordance with the Federal Deposit Insurance Act, and the Bank has paid all premiums and assessments with respect to such deposit insurance.
- 5.17 Reports. Since January 1, 2015, LBI and the Bank have timely filed all reports, registrations and statements, together with any required amendments thereto, that LBI or the Bank was required to file with (i) the Board of Governors of the Federal Reserve System (the "Federal Reserve Board"), (ii) the Indiana Department of Financial Institutions ("IDFI"), (iii) the FDIC, and (iv) any federal, state, municipal or local government, securities, banking, environmental, insurance and other governmental or regulatory authority, and the agencies and staffs thereof (collectively, the "Regulatory Authorities"), having jurisdiction over the affairs of LBI or the Bank, except where such failure would not have a Material Adverse Effect. All such reports filed by LBI and the Bank complied in all material respects with all applicable rules and regulations promulgated by the applicable Regulatory Authorities and were true, accurate and complete in all material respects and, to the extent required, were prepared in conformity with regulatory accounting principles applied on a consistent basis.

- 5.18 <u>Absence of Defaults</u>. Neither LBI nor the Bank is in violation of its respective Articles of Incorporation or By-Laws or, to the knowledge of LBI's Management, in default under any material agreement, commitment, arrangement, loan, lease, insurance policy or other instrument, whether entered into in the ordinary course of business or otherwise and whether written or oral, and there has not occurred any event known to LBI's Management that, with the lapse of time or giving of notice or both, would constitute such a default, except for such defaults which would not have a Material Adverse Effect.
- 5.19 <u>Tax and Regulatory Matters</u>. Neither LBI nor the Bank has taken or agreed to take any action or has any knowledge of any fact or circumstance that would (a) prevent the transactions contemplated hereby from qualifying as a reorganization within the meaning of Section 368 of the Code or (b) materially impede or delay receipt of any regulatory approval required for consummation of the transactions contemplated by this Agreement.

5.20 Real Property.

- (a) A list of the locations of each parcel of real property owned by LBI or the Bank (other than real property acquired in foreclosure or in lieu of foreclosure in the course of the collection of loans and being held by LBI or the Bank for disposition as required by law) is set forth in the LBI Disclosure Letter under the heading of "LBI Owned Real Property" (such real property being herein referred to as the "LBI Owned Real Property"). A list of the locations of each parcel of real property leased by LBI or the Bank is also set forth in the LBI Disclosure Letter under the heading of "LBI Leased Real Property" (such real property being herein referred to as the "LBI Leased Real Property"). LBI shall update the LBI Disclosure Letter within ten (10) days after acquiring or leasing any real property after the date hereof. Collectively, the LBI Owned Real Property and the LBI Leased Real Property are herein referred to as the "LBI Real Property."
 - (b) There is no pending action involving LBI or the Bank as to the title of or the right to use any of the LBI Real Property.
- (c) Other than the LBI Owned Real Property, neither LBI nor the Bank has any interest in any other real property except interests as a mortgagee, and except for any real property acquired in foreclosure or in lieu of foreclosure and being held for disposition as required by law.
- (d) None of the buildings, structures or other improvements located on the LBI Real Property encroaches upon or over any adjoining parcel of real estate or any easement or right-of-way or "setback" line and all such buildings, structures and improvements are located and constructed in conformity with all applicable zoning ordinances and building codes.
- (e) None of the buildings, structures or improvements located on the LBI Real Property are the subject of any official complaint or notice by any governmental authority of violation of any applicable zoning ordinance or building code, and there is no zoning ordinance, building code, use or occupancy restriction or condemnation action or proceeding pending, or, to the best knowledge of LBI's Management, threatened, with

respect to any such building, structure or improvement. The LBI Real Property is in good condition for its intended purpose, ordinary wear and tear excepted, and has been maintained (as to the LBI Leased Real Property, to the extent required to be maintained by LBI or the Bank) in accordance with reasonable and prudent business practices applicable to like facilities. The LBI Real Property has been used and operated in all material respects in compliance with all applicable laws, statutes, rules, regulations and ordinances applicable thereto during LBI's or the Bank's ownership or tenancy of said property.

- (f) Except as set forth on the LBI Disclosure Letter, and as may be reflected in the Financial Information, and except for liens for taxes not yet due and payable or with respect to such easements, liens, defects or encumbrances, real estate taxes and assessments or other monetary obligations such as contributions to an owners' association, as do not individually or in the aggregate materially adversely affect the use or value of the LBI Owned Real Property and which would not have a Material Adverse Effect, LBI and the Bank have, and at the Effective Date will have, good and marketable title to their respective LBI Owned Real Property, free and clear of all liens, mortgages, security interests, encumbrances and restrictions of any kind or character.
- (g) Except as set forth in the LBI Disclosure Letter and to the knowledge of LBI's Management, LBI or the Bank has not caused or allowed the generation, treatment, storage, disposal or release at any LBI Real Property of any Toxic Substance, except in compliance with all applicable federal, state and local laws and regulations and except where such noncompliance would not reasonably be expected to have a Material Adverse Effect. "Toxic Substance" means any hazardous, toxic or dangerous substance, pollutant, waste, gas or material, including, without limitation, petroleum and petroleum products, metals, liquids, semi-solids or solids, that are regulated under any federal, state or local statute, ordinance, rule, regulation or other law pertaining to environmental protection, contamination, quality, waste management or cleanup.
- (h) Except as disclosed in the LBI Disclosure Letter and to the knowledge of LBI's Management, there are no underground storage tanks located on, in or under any LBI Owned Real Property and no such LBI Owned Real Property has previously contained an underground storage tank. Except as set forth in the LBI Disclosure Letter and to the knowledge of LBI's Management, LBI or the Bank do not own or operate any underground storage tank at any LBI Leased Real Property and no such LBI Leased Real Property has previously contained an underground storage tank. To the knowledge of LBI's Management, no LBI Real Property is or has been listed on the Comprehensive Environmental Response, Compensation, and Liability Information System ("CERCLIS").
- (i) Except as set forth in the LBI Disclosure Letter and to the knowledge of LBI's Management, no Toxic Substance has been released, spilled, discharged or disposed at, in, on or under any LBI Real Property nor, to the knowledge of LBI's Management, are there any other conditions or circumstances affecting any LBI Real Property, in each case, which would reasonably be expected to have a Material Adverse Effect.
- (j) To the knowledge of LBI's Management, there are no mechanic's or materialman's liens against the LBI Leased Real Property, and no unpaid claims for labor performed, materials furnished or services rendered in connection with constructing, improving or repairing the LBI Leased Real Property in respect of which liens may or could be filed against the LBI Leased Real Property.

- 5.21 <u>Securities Law Compliance</u>. The LBI Common Stock is not listed on any stock exchange and not actively traded on the over the counter market. LBI has complied in all material respects with all applicable federal and state securities laws, statutes, rules and regulations in regard to the issuance of all share of the LBI Common Stock.
- 5.22 <u>Broker's or Finder's Fees</u>. Except for Renninger & Associates, LLC, no agent, broker or other person acting on behalf of LBI or the Bank or under any authority of LBI or the Bank is or shall be entitled to any commission, broker's or finder's fee or any other form of compensation or payment from any of the parties hereto, other than attorneys' or accountants' fees, in connection with any of the transactions contemplated by this Agreement.
- 5.23 <u>Shareholder Rights Plan</u>. LBI does not have a shareholder rights plan or any other plan, program or agreement involving, restricting, prohibiting or discouraging a change in control or merger of LBI or the Bank or which may be considered an anti-takeover mechanism.
- 5.24 <u>Indemnification Agreements</u>. Except as set forth in the LBI Disclosure Letter, neither LBI nor the Bank is a party to any indemnification, indemnity or reimbursement agreement, contract, commitment or understanding to indemnify any present or former director, officer, employee, shareholder or agent against any liability or hold the same harmless from liability other than as expressly provided in the Articles of Incorporation or By-Laws (or other organizational documents, as applicable) of LBI or the Bank.
- 5.25 <u>Agreements with Regulatory Agencies.</u> Except as set forth in the LBI Disclosure Letter in a manner permitted by applicable law, neither LBI nor the Bank is subject to any cease-and-desist, consent order, or other order or enforcement action issued by, or is a party to any written agreement, consent agreement, or memorandum of understanding with, or is a party to any commitment letter or similar undertaking to, or is subject to any order or directive by, or has been ordered to pay any civil money penalty by, the Regulatory Authorities or other governmental entity.
- 5.26 <u>Nonsurvival of Representations and Warranties</u>. The representations and warranties contained in this Section 5 shall expire on the Effective Date or the earlier termination of this Agreement, and thereafter LBI and the Bank and all directors and officers of LBI and the Bank shall have no further liability with respect thereto.

SECTION 6

REPRESENTATIONS AND WARRANTIES OF F&M

F&M hereby makes the following representations and warranties set forth below to LBI with respect to itself and the F&M Subsidiaries. For the purposes of this Agreement, "**F&M Disclosure Letter**" is defined as a letter referencing Section 6 of this Agreement which shall be prepared by F&M and delivered to LBI contemporaneous with the execution of this Agreement.

6.1 <u>Organization and Qualification</u>. F&M is a corporation duly organized and validly existing under the laws of the State of Ohio and F&M Bank is a commercial bank duly organized and validly existing under the laws of the State of Ohio. F&M and F&M Bank have the power and authority (corporate or otherwise) to conduct their respective businesses in the manner and by the means utilized as of the date hereof. F&M's only subsidiaries are F&M Bank and Farmers & Merchants Risk Management, Inc., a Nevada corporation (the "F&M Subsidiaries"). None of the F&M Subsidiaries have subsidiaries. F&M Bank is subject to primary federal regulatory supervision and regulation by the FDIC.

6.2 Authorization.

- (a) F&M has the corporate power and authority to enter into this Agreement and to carry out their obligations hereunder subject to the conditions precedent set forth in Section 9. The Agreement, when executed and delivered, will have been duly authorized and will constitute a valid and binding obligation of F&M, subject to the conditions precedent set forth in Section 9 hereof, enforceable in accordance with its terms, except to the extent limited by insolvency, reorganization, liquidation, readjustment of debt, or other laws of general application relating to or affecting the enforcement of creditor's rights. The respective Board of Directors of F&M and F&M Bank, and F&M as the sole shareholder of F&M Bank, have approved the Merger and the Bank Merger pursuant to the terms and conditions of this Agreement and the Bank Merger Agreement.
- (b) Except as set forth in the F&M Disclosure Letter, neither the execution of this Agreement, nor the consummation of the transactions contemplated hereby, subject to the conditions precedent set forth in Section 9 hereof does or will (i) conflict with, result in a breach of, or constitute a default under either F&M's or F&M Bank's Articles of Incorporation or Code of Regulations; (ii) conflict with, result in a breach of, or constitute a default under any federal, foreign, state, or local law, statute, ordinance, rule, regulation, or court or administrative order or decree, or any note, bond, indenture, loan, mortgage, security agreement, contract, arrangement, or commitment, to which either F&M or F&M Bank is subject or bound, the result of which would have a Material Adverse Effect; (iii) result in the creation of, or give any person, corporation or entity the right to create, any lien, charge, claim, encumbrance, security interest, or any other rights of others or other adverse interest upon any right, property or asset of either F&M or F&M Bank; (iv) terminate, or give any person, corporation or entity the right to terminate, amend, abandon, or refuse to perform, any note, bond, indenture, loan, mortgage, security agreement, contract, arrangement, or commitment to which F&M or F&M Bank is a party or by which either F&M or F&M Bank is subject or bound, the result of which would have a Material Adverse Effect on F&M; or (v) accelerate or modify, or give any party thereto the right to accelerate or modify, the time within which, or the terms according to which, either F&M or F&M Bank is to perform any duties or obligations or receive any rights or benefits under any note, bond, indenture, loan, mortgage, security agreement, contract, arrangement, or commitment.
- (c) Other than in connection or in compliance with the provisions of the Bank Holding Company Act of 1956, the Bank Merger Act, federal and state securities laws, and applicable federal, Ohio and Indiana banking statutes and corporate statutes, all as amended, and the rules and regulations promulgated thereunder, no notice to, filing with, authorization of, exemption by, or consent or approval of, any public body or authority is necessary for the consummation by F&M and F&M Bank of the transactions contemplated by this Agreement.

(d) Except as set forth in the F&M Disclosure Letter, other than those filings, authorizations, consents and approvals referenced in Section 6.2(c) above and filings and approvals relating to the listing of the shares of F&M Common Stock to be issued in the Merger on the NASDAQ Capital Market and certain other filings and approvals with NASDAQ relating to the change in the number of shares of F&M outstanding as a result of the Merger, no notice to, filing with, authorization of, exemption by, or consent or approval of, any third party is necessary for the consummation by F&M or F&M Bank of the transactions contemplated by this Agreement, except for such authorizations, exemptions, consents or approvals, the failure of which to obtain, would not be reasonably likely to result in a Material Adverse Effect.

6.3 Capitalization.

As of July 20, 2018, F&M had 20,000,000 shares of F&M Common Stock authorized, without par value, of which 9,263,910 shares were issued and outstanding. Such issued and outstanding shares of F&M Common Stock have been duly and validly authorized by all necessary corporate action of F&M, are validly issued, fully paid and nonassessable and have not been issued in violation of any preemptive rights of any shareholders. Other than the F&M Common Stock, F&M has no other class of stock authorized.

- (a) F&M owns all of the issued and outstanding shares of capital stock of each of the F&M Subsidiaries. Such issued and outstanding capital stock has been duly and validly authorized by all necessary corporate action, are validly issued, fully paid and nonassessable, and have not been issued in violation of any preemptive rights of any shareholder. All of the issued and outstanding shares of capital stock of the F&M Subsidiaries are owned by F&M free and clear of all liens, pledges, charges, claims, encumbrances, restrictions, security interests, options and preemptive rights and of all other rights of any other person, corporation or entity with respect thereto.
- (b) The shares of F&M Common Stock to be issued pursuant to the Merger will be duly authorized, fully paid, validly issued and nonassessable and subject to no preemptive rights.
- 6.4 <u>Organizational Documents</u>. The Articles of Incorporation and Code of Regulations of F&M and F&M Bank in force as of the date hereof have been delivered to LBI. The documents delivered by it represent true, accurate and complete copies of the corporate documents of F&M and F&M Bank in effect as of the date of this Agreement.
- 6.5 <u>Compliance with Law.</u> To the knowledge of "F&M Management" (as defined below), except as set forth in the F&M Disclosure Letter, neither F&M nor any F&M Subsidiary has engaged in any activity nor taken or omitted to take any action which has resulted or could reasonably be expected to result, in the violation of any local, state, federal or foreign law, statute, rule, regulation or ordinance or of any order, injunction, judgment or decree of any court or

government agency or body, the violation of which could reasonably be expected to have a Material Adverse Effect. Except as set forth in the F&M Disclosure Letter, F&M and each F&M Subsidiary possess all licenses, franchises, permits and other authorizations necessary for the continued conduct of their respective businesses without material interference or interruption. Neither F&M nor any F&M Subsidiary are subject to any agreement, commitment or understanding with, or order and directive of, any regulatory agency or government authority with respect to the business or operations of F&M or any F&M Subsidiary. Except as set forth in the F&M Disclosure Letter, F&M Bank has not received any notice of enforcement actions since January 1, 2015 from any regulatory agency or government authority relating to its compliance with the Bank Secrecy Act, the Truth-in-Lending Act, the Community Reinvestment Act, the Gramm-Leach-Bliley Act of 1999, the USA Patriot Act, the International Money Laundering Abatement and Financial Anti-Terrorism Act of 2001, the Sarbanes-Oxley Act of 2002, the Dodd-Frank Wall Street Reform and Consumer Protection Act or any laws with respect to the protection of the environment or the rules and regulations promulgated thereunder. Except as set forth in the F&M Disclosure Letter, F&M has not received any notice of enforcement actions since January 1, 2015 from any regulatory agency or government authority relating to its compliance with any securities, tax or employment laws applicable to F&M. F&M Bank received a rating of "satisfactory" or better in its most recent examination or interim review with respect to the Community Reinvestment Act.

- 6.6 Accuracy of Statements. No information which has been or shall be supplied by F&M nor any F&M Subsidiary with respect to its respective businesses, operations and financial condition for inclusion in the proxy statement, registration statement, and regulatory applications relating to the Merger or the Bank Merger contains or shall contain (in the case of information relating to the proxy statement at the time it is mailed and for the regulatory applications and registration statement, and each amendment or supplement thereto, if any, at the time it becomes effective) any untrue statement of a material fact or omits or shall omit to state a material fact necessary to make the statements contained therein not misleading.
- 6.7 <u>Litigation and Pending Proceedings</u>. Except as set forth in the F&M Disclosure Letter, there are no claims of any kind, nor any action, suits, proceedings, arbitrations or investigations pending or, to the knowledge of F&M's Management threatened in any court or before any government agency or body, arbitration panel or otherwise (nor does F&M's Management have any knowledge of a basis for any claim, action, suit, proceeding, arbitration or investigation) which could be reasonably expected to have a Material Adverse Effect. To the knowledge of F&M's Management, there are no material uncured violations, criticisms or exceptions, or violations with respect to which material refunds or restitutions may be required, cited in any report, correspondence or other communication to F&M or any F&M Subsidiary as a result of an examination by any regulatory agency or body.

6.8 Financial Statements.

(a) F&M's consolidated audited balance sheets as of the end of the two (2) fiscal years ended December 31, 2017 and 2016, the unaudited consolidated balance sheet for the six months ended June 30, 2018 and the related consolidated statements of income, shareholders' equity and cash flows for the years or period then ended (hereinafter collectively referred to as the "F&M Financial Information") present fairly the consolidated financial condition or position of F&M as of the respective dates thereof and the consolidated results of operations of F&M for the respective periods covered thereby and have been prepared in conformity with GAAP applied on a consistent basis.

- (b) All loans reflected in the F&M Financial Information and which have been made, extended or acquired since December 31, 2017 (i) have been made for good, valuable and adequate consideration in the ordinary course of business; (ii) constitute the legal, valid and binding obligation of the obligor and any guarantor named therein; (iii) are evidenced by notes, instruments or other evidences of indebtedness which are true, genuine and what they purport to be; and (iv) to the extent that F&M Bank has a security interest in collateral or a mortgage securing such loans, are secured by perfected security interests or mortgages naming F&M Bank as the secured party or mortgagee, except for such unperfected security interests or mortgages naming F&M Bank as secured party or mortgagee which, on an individual loan basis, would not materially adversely affect the value of any such loan and the recovery of payment on any such loan if F&M Bank is not able to enforce any such security interest or mortgage.
- 6.9 Absence of Certain Changes. Except for events and conditions relating to the business and interest rate environment in general, the accrual or payment of Merger-related expenses, or as set forth in the F&M Disclosure Letter, since December 31, 2017, no events have occurred which could reasonably be expected to have a Material Adverse Effect. Except as set forth in the F&M Disclosure Letter, between the period from December 31, 2017 to the date of this Agreement, F&M and each F&M Subsidiary have carried on their respective businesses in the ordinary and usual course consistent with their past practices (excluding the incurrence of reasonable fees and expenses of professional advisors related to this Agreement and the transactions contemplated hereby). Since December 31, 2017, there has not been any declaration, setting aside or payment of any dividend or other distribution (whether in cash, stock or property) with respect to F&M Common Stock (other than normal quarterly cash dividends) or any split, combination or reclassification of any stock of F&M or any F&M Subsidiary or any issuance or the authorization of any issuance of any securities in respect of, or in lieu of, or in substitution for F&M Common Stock.
- 6.10 Taxes, Returns and Reports. F&M and the F&M Subsidiaries have (a) duly filed all federal, state, local and foreign tax returns of every type and kind required to be filed by them as of the date hereof, and each return is true, complete and accurate in all material respects; (b) paid all material taxes, assessments and other governmental charges due and payable or claimed to be due and payable upon them or any of their income, properties or assets; and (c) not requested an extension of time for any such payments (which extension is still in force). Except for taxes not yet due and payable, the reserve for taxes on the F&M Financial Information is adequate to cover all of F&M's and the F&M Subsidiaries' tax liabilities (including, without limitation, income taxes and franchise fees) that may become payable in future years with respect to any transactions consummated prior to December 31, 2017. Neither F&M nor the F&M Subsidiaries have or will have, any liability for taxes of any nature for or with respect to the operation of their business, including the assets of any subsidiary, from December 31, 2017, up to and including the Effective Date, except to the extent reflected on the F&M Financial Information or on financial statements of F&M or any subsidiary subsequent to such date and as set forth in the F&M Disclosure Letter. Neither F&M nor the F&M Subsidiaries have received written notice that it is currently under audit by any state or federal taxing authority. Except as set forth in the F&M Disclosure Letter, none of the federal, state, or local tax returns of F&M or the F&M Subsidiaries have been audited by any taxing authority during the past five (5) years.

- 6.11 <u>Deposit Insurance</u>. The deposits of F&M Bank are insured by the FDIC in accordance with the Federal Deposit Insurance Act, and F&M Bank has paid all premiums and assessments with respect to such deposit insurance.
- 6.12 Reports. Since January 1, 2015, F&M and F&M Bank have timely filed all reports, registrations and statements, together with any required amendments thereto, that they were required to file with (i) the Board of Governors of the Federal Reserve System, (ii) the FDIC, (iii) the Ohio Division of Financial Institutions, and (iv) any federal, state, municipal or local government, securities, banking, environmental, insurance and other governmental or regulatory authority, and the agencies and staffs thereof (collectively, the "F&M Regulatory Authorities"), except where such failure would not have a Material Adverse Effect. All such reports filed by F&M and F&M Bank complied in all material respects with all applicable rules and regulations promulgated by the applicable F&M Regulatory Authorities and were true, accurate and complete in all material respects and, to the extent required, were prepared in conformity with GAAP applied on a consistent basis. There is no unresolved violation with respect to any report or statement filed by, or any examination of F&M or F&M Bank.
- 6.13 <u>Absence of Defaults</u>. Neither F&M nor F&M Bank is in violation of its Articles of Incorporation or Code of Regulations or, to the knowledge of F&M's Management, in default under any material agreement, commitment, arrangement, loan, lease, insurance policy or other instrument, whether entered into in the ordinary course of business or otherwise and whether written or oral, and there has not occurred any event known to F&M's Management that, with the lapse of time or giving of notice or both, would constitute such a default, except for defaults which would not have a Material Adverse Effect.
- 6.14 Tax and Regulatory Matters. Neither F&M nor any F&M Subsidiary has taken or agreed to take any action or has any knowledge of any fact or circumstance that would (a) prevent the transactions contemplated hereby from qualifying as a reorganization within the meaning of Section 368 of the Code or (b) materially impede or delay receipt of any regulatory approval required for consummation of the transactions contemplated by this Agreement.
- 6.15 Securities Law Compliance. F&M Common Stock is traded on the NASDAQ Capital Market under the symbol of "FMAO." F&M has complied in all material respects with all applicable state, federal or foreign securities laws, statutes, rules, regulations or orders, injunctions or decrees of any applicable government agency relating or applicable to F&M Common Stock. F&M has complied in all material respects with all applicable rules, regulations, or orders of any applicable self-regulatory organization (including, but not limited to, NASDAQ) relating or applicable to F&M Common Stock. Since January 1, 2013, F&M has filed all reports and other documents required to be filed by it under the Securities Exchange Act of 1934, as amended, and the Securities Act of 1933, as amended (the "1933 Act"), and the regulations promulgated thereunder, and under any rules and regulations of any applicable self-regulatory organization (including, but not limited to, NASDAQ), including F&M's Annual Report on Form 10-K for the year ended December 31, 2017, copies of which have previously been delivered to LBI. All such filings were true, accurate and complete in all material respects as of the dates of the filings (except for information included therein as of a certain date, which shall have been true and correct as of such date), and no such filings, at the time they were filed, contained any untrue statement of a material fact or omitted to state a material fact necessary in order to make the statements made, at the time and in the light of the circumstances under which they were made, not false or misleading.

- 6.16 <u>Broker's or Finder's Fees</u>. Except for ProBank Austin, no agent, broker or other person acting on behalf of F&M or under any authority of F&M is or shall be entitled to any commission, broker's or finder's fee or any other form of compensation or payment from any of the parties hereto, other than attorneys' or accountants' fees, in connection with any of the transactions contemplated by this Agreement.
- 6.17 <u>Agreements with Regulatory Agencies</u>. Except as set forth in the F&M Disclosure Letter in a manner permitted by applicable law, neither F&M nor the F&M Subsidiaries are subject to any cease-and-desist, consent order, or other order or enforcement action issued by, or is a party to any written agreement, consent agreement, or memorandum of understanding with, or is a party to any commitment letter or similar undertaking to, or is subject to any order or directive by, or has been ordered to pay any civil money penalty by, the F&M Regulatory Authorities or other governmental entity.
- 6.18 <u>Sufficient Funds</u>. F&M has, as of the date hereof, sufficient cash on hand or other sources of immediately available funds to enable F&M to timely pay the cash portion of the Merger Consideration and to satisfy its obligations under and to consummate the transactions contemplated by this Agreement.
- 6.19 <u>No Shareholder Approval</u>. No vote or consent of any of the holders of F&M Common Stock is required by law, contract, or NASDAQ listing requirements for F&M to enter into this Agreement and to consummate the Merger.
- 6.20 <u>Nonsurvival of Representations and Warranties</u>. The representations and warranties contained in this Section 6 shall expire on the Effective Date or the earlier termination of this Agreement, and thereafter F&M and F&M Bank and all directors and officers of F&M and F&M Bank shall have no further liability with respect thereto.

SECTION 7

COVENANTS OF LBI

LBI covenants and agrees with F&M and covenants and agrees to cause the Bank to act, as follows:

7.1 Shareholder Approval.

Following the execution of this Agreement, LBI shall take, in accordance with applicable law and its Articles of Incorporation and Bylaws, all action necessary to convene a meeting of its shareholders as promptly as practicable (and in any event within forty-five (45) days following the time when the Registration Statement becomes effective, subject to extension with the consent of F&M, which shall not unreasonably be withheld, conditioned or delayed) to consider and vote upon the adoption of this Agreement and approval of the transactions contemplated hereby (including the Merger) and any other matter required to be approved by the shareholders of LBI in order to consummate the Merger and the transactions contemplated hereby (including any adjournment or postponement thereof, the "Shareholder Meeting").

Subject to Section 7.5 hereof, LBI shall cooperate with F&M in the preparation of an appropriate proxy statement and other proxy solicitation materials (the "**Proxy Statement**") and use its reasonable best efforts to obtain the requisite vote of LBI's shareholders to consummate the Merger and the other transactions contemplated hereby, and shall ensure that the Shareholder Meeting is called, noticed, convened, held and conducted, and that all proxies solicited by LBI in connection with the Shareholder Meeting are solicited in compliance with the Indiana Business Corporation Law, the Articles of Incorporation and Bylaws of LBI, and all other applicable legal requirements. LBI shall keep F&M updated with respect to the proxy solicitation results in connection with the Shareholder Meeting as reasonably requested by F&M.

Subject to Section 7.5 hereof, LBI's Board of Directors shall recommend that LBI's shareholders vote to adopt this Agreement and approve the transactions contemplated hereby (including the Merger) and any other matters required to be approved by LBI's shareholders for consummation of the Merger and the transactions contemplated hereby.

Subject to Section 7.5 hereof, LBI and the Trustee of the ESOP shall provide each participant or beneficiary in the ESOP with the opportunity to provide the Trustee of the ESOP with written directions as to the manner in which the shares of LBI Common Stock allocated to the account of such participant or beneficiary under the ESOP should be voted by the Trustee at the Shareholder Meeting, as required by the terms of the ESOP and Section 409(e) of the Internal Revenue Code. LBI and the Trustee of the ESOP shall distribute a copy of the Proxy Statement to each participant in the ESOP in advance of the Shareholders Meeting and otherwise comply with all requirements imposed by applicable DOL or IRS guidance regarding pass-through voting for ESOPs.

7.2 Other Approvals. As soon as reasonably practicable following the date hereof, LBI and the Bank shall use their reasonable best efforts to procure upon reasonable terms and conditions any consents, authorizations, approvals, registrations, and certificates from any applicable Regulatory Authorities as may be required by applicable law, and to satisfy all other requirements prescribed by law which are necessary for consummation of the Merger and the Bank Merger on the terms and conditions provided in this Agreement.

7.3 Conduct of Business.

(a) Except as otherwise set forth in the LBI Disclosure Letter or as provided for in this Agreement, on and after the date of this Agreement and until the Effective Date or until this Agreement shall be terminated as herein provided, neither LBI nor the Bank shall, without the prior written consent (which may include consent via electronic mail) of F&M, (i) make any changes in their capital structure, including, but not limited to the redemption of shares of common stock; (ii) authorize an additional class of stock or issue, or authorize the issuance of any capital stock or any options or other instruments convertible into shares of capital stock, except pursuant to the exercise of stock options outstanding as of the date of this Agreement; (iii) declare, distribute or pay any dividends on their common shares, or authorize a stock split, or make any other distribution to their shareholders, except for

LBI's regular quarterly cash dividend or distribution in an amount not to exceed 65% of the net earnings of LBI; provided, however, LBI and F&M shall coordinate LBI's dividend and distribution schedule for the quarter in which Closing occurs so that holders of LBI Common Stock do not receive dividends or distributions on both F&M and LBI Common Stock attributable to the same calendar quarter other than a distribution to pay for the tax liability of the shareholders of LBI; (iv) merge, combine or consolidate with or, other than in the ordinary course of business consistent with past practice (including the sale, transfer or disposal of other real estate owned), sell their assets or any of their securities to any other person, corporation or entity, effect a share exchange or enter into any other transaction not in the ordinary course of business; (v) incur any new liability or obligation, make any new commitment, payment or disbursement, enter into any new contract, agreement, understanding or arrangement or engage in any new transaction, or acquire or dispose of any property, other than other real estate owned, or asset the fair market value of which exceeds Fifty Thousand and 00/100 Dollars (\$50,000.00), in the aggregate, except for payments or disbursements made in the ordinary course of business consistent with past practice, the acquisition or disposition of personal or real property in connection with either foreclosures on mortgages or enforcement of security interests, the origination or sale of loans by the Bank in the ordinary course of business and the creation of deposit liabilities and advances from the Federal Home Loan Bank, in each case, in the ordinary course of business consistent with past practice; (vi) subject any of their properties or assets to a mortgage, lien, claim, charge, option, restriction, security interest or encumbrance, except for such mortgages, liens or other encumbrances incurred in the ordinary course of business consistent with past practice; (vii) promote or increase or decrease the rate of compensation (except for promotions and non-material increases in the ordinary course of business and in accordance with past practices) or enter into any agreement to promote or increase or decrease the rate of compensation of any director, officer or employee of LBI or the Bank; (viii) except as set forth in the LBI Disclosure Letter, as specifically authorized by this Agreement or as required by applicable law, execute, create, institute, modify or amend any pension, retirement, savings, stock purchase, stock bonus, stock ownership, stock option, stock appreciation or depreciation right or profit sharing plans, any employment, deferred compensation, consultant, bonus or collective bargaining agreement, group insurance contract or other incentive, welfare or employee benefit plan or agreement for current or former directors, officers or employees of LBI or the Bank, change the level of benefits or payments under any of the foregoing or increase or decrease any severance or termination pay benefits or any other fringe or employee benefits or pay any bonuses other than as required by law or regulatory authorities; (ix) amend their respective Articles of Incorporation or By-Laws from those in effect on the date of this Agreement; (x) except as set forth in the LBI Disclosure Letter or as specifically authorized by this Agreement, modify, amend or institute new employment policies or practices, or enter into, renew, modify, amend or extend any employment or severance agreements with respect to any present or former directors, officers or employees of LBI or the Bank; (xi) give, dispose, sell, convey, assign, hypothecate, pledge, encumber or otherwise transfer or grant a security interest in any capital stock of the Bank; (xii) fail to make additions to the Bank's reserve for loan losses, or any other reserve account, in the ordinary course of business and in accordance with sound banking practices; (xiii) other than in the ordinary course of business consistent with past practice, incur any indebtedness for borrowed money or assume, guarantee, endorse or otherwise as an accommodation become responsible

liable for the obligations of any other individual, corporation or other entity; (xiv) make any loans or establish or expand any deposit or trust relationship not consistent with prior practice, the policies and procedures of the Bank or Law; and (xv) agree in writing or otherwise to take any of the foregoing actions. The prior consent of F&M for the items listed above may be withheld, conditioned or delayed in its sole discretion.

- (b) LBI and the Bank shall maintain, or cause to be maintained, in full force and effect insurance on its properties and operations and fidelity coverage on its directors, officers and employees in such amounts and with regard to such liabilities and hazards as customarily are maintained by other companies operating similar businesses.
- (c) LBI shall provide and shall cause the Bank to provide F&M and its representatives full access, during normal business hours and on reasonable advance notice to LBI, to further information (to the extent permissible under applicable law) and the Bank's premises for purposes of (i) observing the Bank's business activities and operations and to consult with LBI's officers and employees regarding the same on an ongoing basis to verify compliance by LBI with all terms of this Agreement, and (ii) making all necessary preparations for conversion of the Bank's information technology systems, including, but not limited to, installation of hardware or software devices within the Bank's network to perform system penetration testing or assess previous security breaches. F&M may hire, at its expense, a mutually-agreeable third party consultant to perform cybersecurity system testing and monitoring (based on a mutually-agreeable project scope) in order to confirm that the Bank's technology systems are free of security breaches and, if necessary, provide remediation and notices related thereto. LBI and F&M shall each receive the results of the testing and reasonably coordinate their efforts on any potential remediation and notices. None of the foregoing actions shall unduly interfere with the business operations of LBI or the Bank nor shall such actions be permitted if such access relates to, (i) pending or threatened litigation or investigations if, in the opinion of counsel to LBI, such access would or might adversely affect the confidential nature of, or any privilege relating to, the matters being discussed, or (ii) matters involving an Acquisition Proposal. No investigation pursuant to this Section 7.3 shall affect or be deemed to modify any representation or warranty made in this Agreement by LBI. F&M will use such information as is provided to it by LBI or the Bank, or representatives thereof, solely for the purpose of conducting business, legal and financial reviews of LBI and the Bank and for such other purposes as may be related to this Agreement, and F&M will, and will direct all of its agents, employees and advisors to, maintain the confidentiality of all such information in accordance with the terms of Section 8.4 below. Neither LBI nor the Bank shall be required to provide access to or to disclose information where such access or disclosure would violate or prejudice the rights of its customers, jeopardize the attorney-client privilege of the entity in possession or control of such information or contravene any law, rule, regulation, order, judgment, decree, fiduciary duty or binding agreement entered into prior to the date of this Agreement. The parties will make appropriate and reasonable substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply.

7.4 Preservation of Business. On and after the date of this Agreement and until the Effective Date or until this Agreement is terminated as herein provided, LBI and the Bank shall (a) carry on their business diligently, substantially in the same manner as heretofore conducted, and in the ordinary course of business; (b) use commercially reasonable efforts to preserve their business organizations intact, to keep their present officers and employees and to preserve their present relationship with customers and others having business dealings with them; and (c) not do or fail to do anything which will cause a material breach of, or material default in, any contract, agreement, commitment, obligation, understanding, arrangement, lease or license to which they are a party or by which they are or may be subject or bound.

7.5 Other Negotiations.

- (a) LBI shall not, and shall cause the Bank to not, during the term of this Agreement, directly or indirectly through agents, directors, officers or otherwise, solicit, encourage or facilitate inquiries or proposals or enter into any agreement with respect to, or initiate or participate in any negotiations or discussions with any person or entity concerning, any proposed transaction or series of transactions involving or affecting LBI or the Bank (or the securities or assets of the foregoing) that, if effected, would constitute an acquisition of control of either LBI or the Bank within the meaning of 12 U.S.C. §1817(j) (disregarding the exceptions set forth in 12 U.S.C. §1817(j)(17)) and the regulations of the Federal Reserve Board thereunder (each, an "Acquisition Proposal"), or furnish any information to any person or entity proposing or seeking an Acquisition Proposal.
- (b) Notwithstanding the foregoing, in the event that LBI's Board of Directors determines in good faith and after consultation with outside counsel, that in light of an Acquisition Proposal, it is necessary to provide such information or engage in such negotiations or discussions in order to act in a manner consistent with such Board's fiduciary duties, LBI's Board of Directors may, in response to an Acquisition Proposal which was not solicited by or on behalf of LBI or the Bank or which did not otherwise result from a breach of Section 7.5(a), subject to its compliance with Section 7.5(c), (i) furnish information with respect to LBI or the Bank to such person or entity making such Acquisition Proposal pursuant to a customary confidentiality agreement that is no less restrictive than the Confidentiality Agreement between LBI and F&M, and (ii) participate in discussions or negotiations regarding such Acquisition Proposal. In the event that LBI's Board of Directors determines in good faith and after consultation with outside counsel, that the Acquisition Proposal is a Superior Acquisition Proposal (as defined below) and that it is necessary to pursue such Superior Acquisition Proposal in order to act in a manner consistent with such Board's fiduciary duties, LBI may (A) withdraw, modify or otherwise change in a manner adverse to F&M, the recommendation of LBI's Board of Directors to its shareholders with respect to this Agreement and the Merger, and/or (B) terminate this Agreement in order to concurrently enter into an agreement with respect to such Superior Acquisition Proposal; provided, however, that LBI's Board of Directors may not terminate this Agreement pursuant to this Section 7.5(b) unless and until (x) ten (10) business days have elapsed following the delivery to F&M of a written notice of such determination by LBI's Board of Directors and during such ten (10) business-day period, LBI and the Bank otherwise cooperate with F&M with the intent of enabling the parties to engage in good faith negotiations so that the Merger and other transactions contemplated hereby may be effected and (y) at the end of such ten (10) business-day period LBI's Board of Directors continues reasonably to believe the Acquisition Proposal at issue constitutes a Superior Acquisition Proposal. A "Superior Acquisition Proposal" shall mean any Acquisition

Proposal containing terms which LBI's Board of Directors determines in its good faith judgment (based on the advice of an independent financial advisor) to be more favorable to LBI's shareholders than the Merger and for which financing, to the extent required, is then committed or which, in the good faith judgment of LBI's Board of Directors, is reasonably capable of being obtained by such third party, but shall exclude any Acquisition Proposal the terms of which were made known to LBI's Board of Directors prior to the date of this Agreement.

- (c) In addition to the obligations of LBI set forth in Section 7.5(a) and (b), LBI shall advise F&M orally and in writing as soon as reasonably practicable of any request (whether oral or in writing) for information or of any inquiries, proposals, discussions or indications of interest (whether oral or in writing) with respect to any Acquisition Proposal, the material terms and conditions of such request or Acquisition Proposal and the identity of the person or entity making such request or Acquisition Proposal. LBI shall keep F&M reasonably informed of the status and details (including amendments or proposed amendments) of any such request or Acquisition Proposal, including the status of any discussions or negotiations with respect to any Superior Acquisition Proposal.
- Announcement; Press Releases. In connection with the execution of this Agreement, LBI and F&M intend to jointly issue a press release mutually acceptable to the parties. Except as otherwise required by law, neither LBI nor the Bank shall issue any additional press releases or make any other public announcements or disclosures relating to the Merger and the other transactions contemplated hereby without the prior approval of F&M provided, however, that nothing in this Section 7.6 shall be deemed to prohibit any party from making any disclosure that its counsel deems necessary in order to satisfy such party's disclosure obligation imposed by law.
- 7.7 <u>LBI Disclosure Letter</u>. LBI shall supplement, amend and update as of the Effective Date the LBI Disclosure Letter with respect to any matters hereafter arising which, if in existence or having occurred as of the date of this Agreement, would have been required to be set forth or described in the LBI Disclosure Letter. If, at any time prior to the Effective Date, LBI becomes aware of a fact or matter that might indicate that any of the representations and warranties of LBI herein may be untrue, incorrect or misleading in any material respect, LBI shall promptly disclose such fact or matter to F&M in writing.
- 7.8 <u>Confidentiality.</u> LBI and the Bank shall use commercially reasonable efforts to cause their respective officers, employees, and authorized representatives to hold in strict confidence all confidential data and information obtained by them from F&M, unless such information (a) was already known to LBI and the Bank, (b) becomes available to LBI and the Bank from other sources, (c) is independently developed by LBI and the Bank, (d) is disclosed by LBI or the Bank with and in accordance with the terms of prior written approval of F&M, or (e) is or becomes readily ascertainable from public or published information or trade sources or public disclosure of such information is required by law or requested by a court or other governmental agency, commission, or regulatory body. LBI and the Bank further agree that, in the event this Agreement is terminated, they will return to F&M, or destroy, all information obtained by LBI and the Bank from F&M Bank, including all copies made of such information by LBI and the Bank. This provision shall survive the Effective Date or the earlier termination of this Agreement.

- 7.9 <u>Cooperation</u>. LBI and the Bank shall generally cooperate with F&M and its officers, employees, attorneys, accountants and other agents, and, generally, do such other acts and things in good faith as may be reasonable, necessary or appropriate to timely effectuate the intents and purposes of this Agreement and the consummation of the transactions contemplated hereby, including, without limitation, (a) LBI shall cooperate and assist F&M in the preparation of and/or filing of all regulatory applications, the Registration Statement, and all other documentation required to be prepared for consummation of the Merger and the Bank Merger and obtaining all necessary approvals, and (b) LBI shall furnish F&M with all information concerning itself and the Bank that F&M may request in connection with the preparation of the documentation referenced above.
- 7.10 <u>LBI Fairness Opinion</u>. On or prior to the date hereof Renninger & Associates, LLC has delivered its oral opinion to the LBI Board of Directors to the effect that the Merger Consideration is fair, from a financial point of view to the holders of the LBI Common Stock. On the date hereof or as soon as reasonably practicable following the date hereof (but in no event more than three (3) days), LBI shall use its reasonable best efforts to procure the written opinion from Renninger & Associates, LLC to the Board of Directors of LBI to the effect that, as of the date of this Agreement, the Merger Consideration is fair, from a financial point of view, to the holders of LBI Common Stock (the "LBI Fairness Opinion"). The LBI Fairness Opinion shall be included in the Proxy Statement (as defined below).
- 7.11 <u>Financial Statements and Other Reports</u>. Promptly upon its becoming available, LBI shall furnish to F&M one (1) copy of each financial statement, report, notice, or proxy statement sent by LBI to its shareholders generally or filed with any Regulatory Authorities.
- 7.12 Adverse Actions. Neither LBI nor the Bank shall (a) take any action while knowing that such action would, or is reasonably likely to, prevent or impede the Merger from qualifying as a reorganization within the meaning of Section 368 of the Code; or (b) knowingly take any action that is intended or is reasonably likely to result in (i) any of its representations and warranties set forth in this Agreement being or becoming untrue in any material respect at any time at or prior to the Effective Date, (ii) any of the conditions to the Merger set forth in Section 9 not being satisfied, (iii) a material violation of any provision of this Agreement, or (iv) a material delay in the consummation of the Merger except, in each case, as may be required by applicable law or regulation.
- 7.13 <u>Bank Merger Agreement.</u> LBI shall cause the appropriate officers of the Bank to execute and deliver the Bank Merger Agreement contemporaneously herewith.
- 7.14 Employment Agreement and Change in Control Agreement. F&M has entered into an employment agreement with Phillip Lucas as of the date hereof that will become effective as of the Effective Date (the "Employment Agreement"). A copy of the Employment Agreement has been provided to LBI. LBI will cause the termination of the LBI change in control agreement with Phillip Lucas as more fully described in the LBI Disclosure Letter (the "Change in Control Agreement") and pay the amounts due thereunder on the Effective Date. As of the date hereof, Phillip Lucas has agreed to termination of the Change in Control Agreement upon payment of the amounts set forth therein.

7.15 <u>ESOP and 401(k) Profit Sharing Plan</u>. LBI shall cause the ESOP and the LBI 401(k) Profit Sharing Plan to be terminated effective prior to the Effective Date.

SECTION 8

COVENANTS OF F&M

F&M covenants and agrees with LBI as follows:

8.1 Approvals. As soon as reasonably practicable, but in any event within sixty (60) days following execution and delivery of this Agreement, F&M will file an application with the Federal Reserve Board (or seek a waiver of such filing obligation) and the Ohio Division of Financial Institutions ("ODFI") and an application with the ODFI and the FDIC for approval of and provide notice to the IDFI with respect to the Bank Merger, and take all other appropriate actions necessary to obtain the regulatory approvals referred to herein, and LBI will use all reasonable and diligent efforts to assist in obtaining all such approvals. In advance of filing any applications for such regulatory approvals, F&M shall provide LBI and its counsel with a copy of such applications (but excluding any information contained therein regarding F&M and its business or operations for which confidential treatment will be requested) and provide an opportunity to comment thereon, and thereafter shall promptly advise LBI and its counsel of any material communication received by F&M or its counsel from any regulatory authorities with respect to such applications. In addition, F&M agrees to prepare, in cooperation with and subject to the review and comment of LBI and its counsel, a registration statement on Form S-4, including a prospectus of F&M (the "Registration Statement"), to be filed no later than sixty (60) days after the date hereof by F&M with the SEC in connection with the issuance of F&M Common Stock in the Merger (including the proxy statements and prospectus and other proxy solicitation materials of, and to be filed by, LBI and F&M constituting a part thereof (the "Proxy Statement") and all related documents). F&M agrees to use its reasonable best efforts to have the Registration Statement declared effective by the SEC and to keep the Registration Statement effective so long as is necessary to consummate the Merger and the transactions contemplated hereby. F&M agrees to advise LBI, promptly after F&M receives notice thereof, of the time when the Registration Statement has become effective or any supplement or amendment has been filed, of the issuance of any stop order or the suspension of the qualification of F&M Common Stock for offering or sale in any jurisdiction, of the initiation or threat of any proceeding for any such purpose, or of the receipt of any comment letters from the SEC regarding, or of any request by the SEC for the amendment or supplement of, the Registration Statement, or for additional information, F&M agrees to use its reasonable best efforts to list, prior to the Effective Date, on the NASDAQ Capital Market (subject to official notice of issuance), the shares of F&M Common Stock to be issued to the holders of shares of LBI Common Stock in the Merger.

8.2 Employee Benefit Plans.

(a) F&M shall take such action as may be necessary so that, as soon as reasonably practicable following the Effective Date, employees of LBI and the Bank shall be entitled to participate in the employee benefit plans of F&M and F&M Bank. Notwithstanding the preceding sentence, at its discretion, F&M may either (a) assume sponsorship of the Bank's group health plan, continue to maintain such group health plan and provide employees of LBI and the Bank with coverage under such LBI group health

plan through the end of the current policy year which includes the Effective Date, subject to and pursuant to the terms thereof; or (b) direct LBI and the Bank to take such steps to terminate such LBI group health plan on the Effective Date and then enroll employees of LBI and the Bank in a group health plan maintained by F&M and F&M Bank effective immediately on the Effective Date.

- (b) With respect to each employee benefit plan or benefit arrangement maintained by F&M or F&M Bank in which employees of LBI or the Bank subsequently participate, for purposes of determining eligibility, vesting, vacation and severance entitlement, F&M will ensure, or cause F&M Bank to ensure, that service with LBI or the Bank will be treated as service with F&M or F&M Bank; provided, however, that service with LBI or the Bank shall not be treated as service with F&M for purposes of benefit accrual, except with respect to vacation and severance benefits. F&M and F&M Bank shall honor all benefits vested and earned by employees of LBI and the Bank through the Effective Date. Upon merger or termination of the Employee Plans or other employee benefit plans of LBI or the Bank, F&M and F&M Bank shall honor any acceleration provision therein.
- (c) With respect to F&M and F&M Bank group health plans under which employees of LBI or the Bank and their eligible dependents become participants, F&M agrees, and to cause F&M Bank to agree, to (i) waive all restrictions and limitations for pre-existing conditions and (ii) waive any waiting period limitation or evidence of insurability requirement that would otherwise be applicable to an employee of LBI or the Bank and his or her eligible dependents on or after the Effective Date, in each case to the extent such employee or eligible dependent had satisfied any similar limitation or requirement under an analogous plan prior to the Effective Date. To the extent employees of LBI and the Bank are required to enroll in a F&M or F&M Bank group health plan prior to the end of the current plan year of LBI's group health plan, F&M agrees, and to cause F&M Bank to agree, to use reasonable efforts to ensure that its group health plan will honor any deductible, co-payments and out-of-pocket maximums incurred by LBI's or the Bank's employees and their eligible dependents under the health plans in which they participated immediately prior to the Effective Date during the portion of the calendar year prior to the Effective Date in satisfying any deductibles, co-payments or out-of-pocket maximums under health plans of F&M or F&M Bank in which they are eligible to participate after the Effective Date in the same plan year in which such deductibles, co-payments or out-of-pocket maximums were incurred.
- (d) F&M agrees, and to cause F&M Bank to agree, that the employees of LBI and the Bank will be permitted to elect to make rollover contributions to the 401(k) plan maintained by F&M and/or F&M Bank as direct rollovers from the terminated LBI 401(k) Plan and ESOP, provided that (i) F&M may, in its discretion, make such direct rollovers subject to and conditioned upon the receipt of IRS determination letters on the termination of the LBI 401(k) plan and the ESOP, and (ii) such rollover contributions will be completed within the period of ninety (90) days following receipt of such IRS determination letters.

- (e) Except for Phillip Lucas, and any other employee receiving a separate change in control, severance or similar payment in connection with the Closing of the Merger, those employees of LBI or the Bank as of the Effective Time (i) who are still employed by the Bank and who F&M Bank elects not to employ after the Effective Time or who are terminated other than for cause within six (6) months after the Effective Date; and (ii) who sign and deliver a termination and release agreement in a form acceptable to F&M, shall be entitled to severance pay equal to two (2) weeks of pay, at their base rate of pay in effect at the time of termination, for each full year of continuous service with LBI and the Bank with a minimum of four (4) weeks and a maximum of twenty-six (26) weeks. Nothing in this Section shall be deemed to limit or modify F&M's or F&M Bank's at-will employment policy or any employee's at will employment status.
- (f) F&M shall be responsible for providing COBRA continuation coverage to any qualified employee or former employee of LBI or the Bank and to their respective qualified beneficiaries, on and after the Effective Date, regardless of when the qualifying event occurred.
- (g) Contemporaneously with the execution of this Agreement, F&M shall enter into the Employment Agreement, effective upon the Effective Date, with Phillip Lucas.
- 8.3 <u>Announcement; Press Releases</u>. In connection with the execution of this Agreement, LBI and F&M intend to jointly issue a press release mutually acceptable to the parties. Except as otherwise required by law, neither F&M nor F&M Bank shall issue any additional press releases or make any other public announcements or disclosures relating to the Merger or the Bank Merger without the prior approval of LBI provided, however, that nothing in this Section 8.3 shall be deemed to prohibit any party from making any disclosure that its counsel deems necessary in order to satisfy such party's disclosure obligation imposed by law.
- 8.4 <u>Confidentiality.</u> F&M shall, and shall use its best efforts to cause the F&M Bank and its officers, employees, and authorized representatives to, hold in strict confidence all confidential data and information obtained by them from LBI or the Bank, unless such information (i) was already known to F&M prior to entering into merger discussions with LBI, (ii) becomes available to F&M from other sources, (iii) is independently developed by F&M, (iv) is disclosed by F&M with and in accordance with the terms of prior written approval of LBI, or (v) is or becomes readily ascertainable from public or published information or trade sources or public disclosure of such information is required by law or requested by a court or other governmental agency, commission, or regulatory body. F&M further agrees that in the event this Agreement is terminated, it will return to LBI, or will destroy, all information obtained by it regarding LBI or the Bank, including all copies made of such information by F&M. This provision shall survive the Effective Date or the earlier termination of this Agreement.

8.5 <u>Directors and Officers Insurance</u>.

(a) For a period of at least six (6) years from the Effective Date (the "Tail Coverage Period"), F&M shall use its reasonable best efforts to obtain an endorsement to its director's and officer's liability insurance policy to cover the present and former officers and directors of LBI and the Bank (determined as of the Effective Date) with respect to claims against such directors and officers arising from facts or events which occurred before the Effective Date, which insurance shall contain at least the same coverage and amounts, and contain terms and conditions no less advantageous, as that coverage currently provided by LBI; provided, however, that if F&M is unable to obtain such endorsement,

then F&M may purchase tail coverage under LBI's existing director and officer liability insurance policy for such claims; provided further that in no event shall F&M be required to expend in the aggregate during each year of the Tail Coverage Period more than 1.5 times the annual amount paid by LBI for its director and officer liability insurance coverage prior to the Effective Date (the "Insurance Amount"); provided further, that if F&M is unable to maintain or obtain the insurance called for by this Section 8.5, F&M shall use its reasonable best efforts to obtain as much comparable insurance as is available for the Insurance Amount; provided, further, that officers and directors of LBI or the Bank may be required to make application and provide customary representations and warranties to F&M' insurance carrier for the purpose of obtaining such insurance.

- (b) Following the Effective Date, F&M will provide any LBI or Bank officers, directors and employees who become officers, directors and employees of the Continuing Company or its subsidiaries with the same directors and officers liability insurance coverage and indemnification protections that F&M provides to other officers, directors and employees of F&M or its subsidiaries. In addition, F&M further agrees to indemnify and advance expenses to the current and former directors and officers of LBI and the Bank after the Effective Date, for all actions taken by them prior to the Effective Date in their respective capacities as directors and officers of LBI or the Bank to the same extent (and subject to the same limitations) as the indemnification provided by LBI and the Bank under their respective Articles of Incorporation and By-laws (as applicable) to such directors and officers immediately prior to the Effective Date and as permitted under applicable law. Notwithstanding the foregoing, the indemnity obligations contained herein shall be limited as may be required by applicable federal banking laws and regulations.
- (c) All rights to indemnification and exculpation from liabilities for acts or omissions occurring on or prior to the Effective Date now existing in favor of the current or former directors or officers of LBI and the Bank as provided in their respective Articles of Incorporation and By-laws and any existing indemnification agreements or arrangements of LBI or the Bank described in the LBI Disclosure Letter, shall survive the Merger and Bank Merger and shall continue in full force and effect in accordance with their terms to the extent permitted by law, and shall be honored by F&M following the Effective Date with respect to acts or omissions of such individuals occurring or alleged to occur on or prior to the Effective Date.
- (d) In the event of any threatened or actual claim, action, suit, proceeding or investigation, whether civil, criminal or administrative, including, without limitation, any such claim, action, suit, proceeding or investigation in which any individual who is now, or has been at any time prior to the date of this Agreement, or who becomes prior to the Effective Date, a director or officer of LBI or the Bank (the "Indemnified Parties"), is, or is threatened to be, made a party based in whole or in part on, or arising in whole or in part out of, or pertaining to (i) the fact that he is or was a director, officer or employee of LBI or the Bank or any of their predecessors or (ii) this Agreement or any of the transactions contemplated hereby, whether in any case asserted or arising before or on or after the Effective Date, the parties hereto agree to cooperate and use their best reasonable efforts to defend against and respond thereto.

- (e) If F&M shall consolidate with or merge into any other entity and shall not be the continuing or surviving entity of such consolidation or merger or shall transfer all or substantially all of its assets to any entity, then and in each case, proper provision shall be made so that the successors and assigns of F&M shall assume the obligations set forth in this Section 8.5.
- 8.6 <u>SEC and Other Reports</u>. Promptly upon its becoming available, F&M shall furnish to LBI one (1) copy of each financial statement, report, notice, or proxy statement sent by F&M to its shareholders generally and of each regular or periodic report, registration statement or prospectus filed by F&M with the SEC or any successor agency, and of any notice or communication received by F&M from the SEC, which is not available on the SEC's EDGAR internet database.
- 8.7 <u>F&M Disclosure Letter</u>. F&M shall supplement, amend and update as of the Effective Date the F&M Disclosure Letter with respect to any matters hereafter arising which, if in existence or having occurred as of the date of this Agreement, would have been required to be set forth or described in the F&M Disclosure Letter. If, at any time prior to the Effective Date, F&M becomes aware of a fact or matter that might indicate that any of the representations and warranties of F&M herein may be untrue, incorrect or misleading in any material respect, F&M shall promptly disclose such fact or matter to LBI in writing.
- 8.8 Adverse Actions. Neither F&M nor F&M Bank shall (a) take any action while knowing that such action would, or is reasonably likely to, prevent or impede the Merger from qualifying as a reorganization within the meaning of Section 368 of the Code; or (b) knowingly take any action that is intended or is reasonably likely to result in (i) any of its representations and warranties set forth in this Agreement being or becoming untrue in any respect at any time at or prior to the Effective Date, (ii) any of the conditions to the Merger set forth in Section 9 not being satisfied, (iii) a material violation of any provision of this Agreement, or (iv) a material delay in the consummation of the Merger except, in each case, as may be required by applicable law or regulation.
- 8.9 <u>Cooperation</u>. F&M shall generally cooperate with LBI and its officers, employees, attorneys, accountants and other agents, and, generally, do such other acts and things in good faith as may be reasonable, necessary or appropriate to timely effectuate the intents and purposes of this Agreement and the consummation of the transactions contemplated hereby.
- 8.10 <u>Bank Merger Agreement.</u> F&M shall cause the appropriate officers of F&M Bank to execute and deliver the Bank Merger Agreement upon approval by F&M Bank's Board of Directors.
- 8.11 <u>Preservation of Business</u>. On and after the date of this Agreement and until the Effective Date or until this Agreement is terminated as herein provided, F&M and F&M Bank shall (a) except as set forth in the F&M Disclosure Letter, carry on their business diligently, substantially in the same manner as heretofore conducted, and in the ordinary course of business; (b) use commercially reasonable efforts to preserve their business organizations intact, to keep their present officers and employees and to preserve their present relationship with customers and others having business dealings with them; and (c) not do or fail to do anything which will cause a material breach of, or material default in, any contract, agreement, commitment, obligation, understanding, arrangement, lease or license to which they are a party or by which they are or may be subject or bound.

8.12 <u>Representation on F&M and F&M Bank Boards</u>. F&M shall cause Andrew J. Briggs, currently the Chairman of LBI and President of the Bank, to be appointed to the F&M and F&M Bank Boards of Directors according to the F&M and F&M Bank Articles of Incorporation, Code of Regulations, and other corporate governance guidelines and requirements, effective as of the Effective Date.

SECTION 9

CONDITIONS PRECEDENT TO THE MERGER AND THE BANK MERGER

The obligation of each of the parties hereto to consummate the transactions contemplated by this Agreement is subject to the satisfaction and fulfillment of each of the following conditions on or prior to the Effective Date:

- 9.1 Shareholder Approval. The shareholders of LBI shall have approved the Merger as required by applicable law.
- 9.2 <u>Registration Statement Effective</u>. F&M shall have registered its shares of F&M Common Stock to be issued to shareholders of LBI in accordance with this Agreement with the SEC pursuant to the 1933 Act, and all state securities and "blue sky" approvals and authorizations required to offer and sell such shares, if any, shall have been received by F&M. The Registration Statement shall have been declared effective by the SEC and no stop order shall have been issued or threatened. The shares of F&M Common Stock to be issued to shareholders of LBI in accordance with this Agreement shall have been listed for trading on the NASDAQ Capital Market (subject to official notice of issuance).

9.3 Tax Opinions.

- (a) F&M shall have obtained an opinion of Shumaker, Loop & Kendrick, LLP, in form and substance reasonably acceptable to the parties, dated on or about the Effective Date, to the effect that the Merger effected pursuant to this Agreement shall constitute a reorganization within the meaning of Section 368(a) of the Code. Such opinion shall be based upon factual representations received by counsel from LBI and F&M, which representations may take the form of written certifications.
- (b) LBI shall have received a letter from Shumaker, Loop & Kendrick, LLP-addressed to the shareholders of LBI, in form and substance reasonably acceptable to the parties, dated on or about the Effective Date, to the effect that such shareholders shall be permitted to rely on the opinion referred to in Section 9.3(a) above.
- 9.4 <u>Regulatory Approvals</u>. The ODFI shall have authorized and approved the Merger and the transactions related thereto and the Federal Reserve Board shall either have authorized and approved the Merger and the transactions related thereto, or issued a waiver of its approval for the Merger. The ODFI and FDIC shall have approved the Bank Merger and the transactions related thereto. In addition, all appropriate orders, consents, approvals and clearances from all other regulatory agencies and governmental authorities whose orders, consents, approvals or clearances

are required by law for consummation of the transactions contemplated by this Agreement shall have been obtained. All regulatory approvals remain in full force and effect and all statutory waiting periods shall have expired or been terminated. Copies of all such regulatory approvals shall have been delivered to LBI.

- 9.5 Officer's Certificate. F&M and LBI shall have delivered to each other a certificate signed by their respective Chairman or President and their Secretary, dated the Effective Date, certifying that (a) all of the representations and warranties of their respective corporations are true, accurate and correct in all material respects on and as of the Effective Date, except that representations and warranties that are qualified by materiality or a Material Adverse Effect shall be true and correct in all respects, and provided that for those representations and warranties which address matters only as of an earlier date, then they shall be tested as of such earlier date; (b) all the covenants of their respective corporations have been complied with in all material respects from the date of this Agreement through and as of the Effective Date; and (c) their respective corporations have satisfied and fully complied with in all material respects all conditions necessary to make this Agreement effective as to them. Additionally LBI shall certify as to the number of shares of its capital stock are issued and outstanding as of the Effective Date.
- 9.6 <u>Secretary's Certificate</u>. F&M and LBI shall have delivered to each other copies of the resolutions adopted by the Board of Directors of F&M, F&M Bank, LBI, and the Bank, certified by each entity's Secretary as of the Effective Date, relative to the approval of this Agreement and the Merger and the Bank Merger Agreement and the Bank Merger, as applicable.
- 9.7 <u>No Judicial Prohibition</u>. None of LBI, the Bank, F&M or F&M Bank shall be subject to any order, decree or injunction of a court or agency of competent jurisdiction which enjoins or prohibits the consummation of the Merger or the Bank Merger.
- 9.8 <u>LBI Fairness Opinion</u>. LBI shall have obtained the LBI Fairness Opinion. Such opinion shall be provided orally to LBI Board of Directors on or prior to the date hereof and a written copy of such fairness opinion shall be delivered to LBI within three (3) days of the date hereof.
 - 9.9 Bank Merger Agreement. F&M Bank and the Bank shall have entered into the Bank Merger Agreement.
- 9.10 <u>Employment Agreement</u>. F&M Bank shall have entered into a mutually agreed employment agreement with Phillip Lucas providing for employment of Mr. Lucas by F&M Bank immediately after the Effective Date.
- 9.11 <u>Appointment to Board of Directors</u>. F&M and F&M Bank shall have taken all appropriate corporate action to cause, effective as of the Effective Date, Andrew J. Briggs to be appointed to the F&M and F&M Bank Boards of Directors. F&M shall have delivered to LBI a copy of the resolutions adopted by its and F&M Bank's Board of Directors, certified by each entity's Secretary as of the Effective Date, relative to the appointment of Andrew J. Briggs.

- 9.12 Exchange Fund. F&M shall have (i) authorized the issuance of and shall have made available to the Exchange Agent, for the benefit of the registered shareholders of LBI Common Stock for exchange in accordance with Section 3, the F&M stock certificates to be issued pursuant to Section 3.1, or confirmation regarding issuance of the interest in the Common Stock in book entry form, and (ii) deposited with the Exchange Agent sufficient cash for payment of the cash portion of the Merger Consideration and cash in lieu of any fractional shares of F&M Common Stock in accordance with Sections 3.1 and 3.2.
- 9.13 <u>D&O Tail Insurance</u>. F&M shall have delivered to LBI evidence of the endorsement to its director and officer liability insurance policy or purchase of a director and officer liability insurance in accordance with Section 8.5, as applicable.
- 9.14 <u>Miscellaneous</u>. F&M and LBI shall have delivered to each other such other documents and information, in a form reasonable satisfactory to the receiving party, as the other party or its legal counsel may have reasonably requested.

SECTION 10

TERMINATION OF MERGER

- 10.1 <u>Manner of Termination</u>. This Agreement and the transactions contemplated hereby may be terminated at any time prior to the Effective Date by written notice delivered by F&M to LBI or by LBI to F&M only for the following reasons:
 - (a) By the mutual consent of F&M and LBI, if the Board of Directors of each so determines by vote of a majority of the members of its entire Board:
 - (b) By F&M or LBI, if its respective Board of Directors so determines by vote of a majority of the members of its entire Board, in the event of either: (i) a material breach by the other party of any representation or warranty contained herein which breach cannot be or has not been cured within thirty (30) days after the giving of written notice to the breaching party of such breach; (ii) a material breach by the other party of any of the covenants or agreements contained herein, which breach cannot be or has not been cured within thirty (30) days after the giving of written notice to the breaching party of such breach; or (iii) any event, fact or circumstance shall have occurred with respect to the other party that has had or could be reasonably expected to have a Material Adverse Effect on such party;
 - (c) by either F&M or LBI, in the event of the failure of LBI's shareholders to approve the Agreement at the Shareholder Meeting; provided, however, that LBI shall only be entitled to terminate the Agreement pursuant to this clause if it has complied in all material respects with its obligations under Section 7.1;
 - (d) by either F&M or LBI, if either (i) any approval, consent or waiver of any governmental or regulatory authority, agency, court, commission, or other administrative entity ("Governmental Entity") required to permit consummation of the transactions contemplated by this Agreement shall have been denied and such denial has become final and non-appealable or (ii) any court or other Governmental Entity of competent jurisdiction shall have issued a final, unappealable order enjoining or otherwise prohibiting consummation of the transactions contemplated by this Agreement;

- (e) By LBI or F&M, if the transaction contemplated herein has not been consummated by March 31, 2019; provided that the terminating party is not then in material breach of any representation, warranty, covenant or other agreement contained herein;
 - (f) By LBI, in accordance with the terms of Section 7.5(b) of this Agreement;
- (g) By F&M, if LBI's Board of Directors fails to make, withdraws or modifies its recommendation for LBI's shareholders to vote in favor of the Merger following receipt of a written proposal for an Acquisition Proposal;
- (h) By F&M, (i) if LBI breaches in any material respect its notice obligations under Section 7.5(c) or (ii) if within sixty (60) days after giving F&M written notice pursuant to Section 7.5(c) of an Acquisition Proposal, LBI does not terminate all discussions, negotiations and information exchanges related to such Acquisition Proposal and provide F&M with written notice of such termination; or
- (i) By F&M if greater than ten percent (10%) of the outstanding shares of LBI Common Stock have become and remain Dissenting Shares as described in Section 4.
- 10.2 Effect of Termination. Except as provided below, in the event that this Agreement is terminated pursuant to the provisions of Section 10.1 hereof, this Agreement shall forthwith become void and, no party shall have any liability to any other party for costs, expenses, damages or otherwise, except that Sections 7.8, 8.4, 10.2, 13.9, and 13.12 shall survive any termination of this Agreement; provided, however, that notwithstanding the foregoing, in the event that this Agreement is terminated pursuant to Section 10.1(b)(i) and (ii) hereof on account of a willful breach of any of the representations and warranties set forth herein or any willful breach of any of the agreements set forth herein, then the non-breaching party shall be entitled to recover appropriate damages from the breaching party, including, without limitation, reimbursement to the non-breaching party of its costs, fees and expenses (including attorneys', accountants' and advisors' fees and expenses) incident to the negotiation, preparation and execution of this Agreement and related documentation; provided further, however, that nothing in the foregoing proviso shall be deemed to constitute liquidated damages for the breach by a party of the terms of this Agreement or otherwise limit the rights of the non-breaching party. Notwithstanding the foregoing, in the event of termination by LBI in accordance with Section 10.1(f) or by F&M in accordance with Section 10.1(g), then LBI shall pay F&M the sum of Three Million Five Hundred Thousand and 00/100 Dollars (\$3,500,000.00) as a termination fee. Such payment shall be made within ten (10) days of the date of notice of termination. F&M shall also be entitled to recover from LBI its reasonable attorneys' fees incurred in the enforcement of the two (2) immediately preceding sentences. The termination fee payable by the LBI constitutes liquidated damages and not a penalty and shall be the sole remedy of F&M in the event of termination of this Agreement based on Sections 10.1(f) or 10.1(g).

SECTION 11

EFFECTIVE DATE OF MERGER

Subject to the terms and upon satisfaction of all requirements of law and the conditions specified in this Agreement, the Merger shall become effective on the date and at the time specified in the Certificate of Merger (the "Certificate of Merger") as filed with the Secretary of State of the State of Ohio and the Articles of Merger (the "Articles of Merger") as filed with the Secretary of State of the State of Indiana providing for the Merger of LBI with and into F&M (the "Effective Date"). Unless otherwise agreed to by the parties, the Effective Date shall be no later than the last business day of the month in which both (a) any waiting period following the last approval of the Merger and Bank Merger by a state or federal regulatory agency or governmental authority expires and (b) the conditions precedent to the Merger and the Bank Merger outlined in Section 9 have been satisfied, provided that, unless otherwise agreed by the parties, in no event shall the Effective Date be before December 31, 2018.

SECTION 12

CLOSING

- 12.1 <u>Closing Date and Place</u>. The closing of the Merger (the "**Closing**") and the Bank Merger shall take place at the main office of F&M on the Effective Date or at such other time and place as mutually agreed to by F&M and LBI.
- 12.2 <u>Merger Certificate of Merger and Articles of Merger</u>. Subject to the provisions of this Agreement, on or before (provided an effective date and time is stated therein) the Effective Date, the Certificate of Merger shall be duly filed with the Secretary of State of the State of Ohio and the Articles of Merger shall be duly filed with the Secretary of State of Indiana.
- 12.3 <u>Bank Merger Certificate of Merger and Articles of Merger</u>. Subject to the provisions of this Agreement, on or before (provided an effective date and time is stated therein) the Effective Date, the Certificate of Merger shall be duly filed with the Secretary of State of the State of Ohio and the Articles of Merger shall be duly filed with the Secretary of State of the State of Indiana and such other filings as may be necessary to consummate the Bank Merger shall be duly filed.

SECTION 13

MISCELLANEOUS

13.1 <u>Effective Agreement</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but none of the provisions hereof shall inure to the benefit of any other person, firm, or corporation whomsoever; except that (a) the terms and provisions of Sections 8.2(e) and 8.5 of this Agreement shall inure to the benefit of the current and former employees, officers and directors of LBI, as applicable, as specified in such sections and shall be enforceable by such individuals against F&M, and (b) the terms and provisions of Section 3.4 shall inure to the benefit of the former shareholders of LBI. Neither this Agreement nor any of the rights, interests, or obligations hereunder shall be assigned or transferred by either party hereto without the prior written consent of the other party.

13.2 Waiver; Amendment.

- (a) F&M and LBI may, by an instrument in writing executed in the same manner as this Agreement: (i) extend the time for the performance of any of the covenants or agreements of the other party under this Agreement; (ii) waive any inaccuracies in the representations or warranties of the other party contained in this Agreement or in any document delivered pursuant hereto or thereto; (iii) waive the performance by the other party of any of the covenants or agreements to be performed by it or them under this Agreement; or (iv) waive the satisfaction or fulfillment of any condition the nonsatisfaction or nonfulfillment of which is a condition to the right of the party so waiving to terminate this Agreement. The waiver by any party hereto of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any other or subsequent breach hereunder.
- (b) Notwithstanding the prior approval by the shareholders of LBI, this Agreement may be amended, modified or supplemented by the written agreement of LBI and F&M without further approval of such shareholders, except that no such amendment, modification or supplement shall decrease the consideration specified in Section 3 hereof, or shall otherwise materially adversely affect the rights of the shareholders of LBI or the tax consequences of the Merger to the shareholders of LBI without the further approval of such shareholders.
- 13.3 <u>Notices</u>. Any and all notices or other communications required or permitted under this Agreement shall be in writing and shall be deemed to be given (i) when delivered in person, (ii) on the next business day, if sent by a nationally recognized overnight courier for overnight delivery (receipt requested), (iii) on the day of transmission if sent via facsimile transmission or electronic mail during normal business hours to the facsimile numbers or email addresses given below, provided telephonic confirmation of receipt is obtained promptly after completion of transmission, or (iv) on the fifth (5th) day after sent by certified or registered mail, postage prepaid, return receipt requested, addressed as follows:

If to F&M:

307 N. Defiance Street Archbold, Ohio 43502 Attn: Paul S. Siebenmorgen

President

FAX: (419) 446-2982

Email: PSiebenmorgen@fm.bank

If to LBI:

215 East Line St Geneva, Indiana 46740-0278 Attn: Andrew J. Briggs Chairman FAX: n/a Email

AB riggs@bank of geneva.com

With a copy to:

Shumaker, Loop & Kendrick, LLP 1000 Jackson Street
Toledo, Ohio 43604-5573
Attn: Thomas C. Blank
FAX: (419) 241-6894
Email: tblank@slk-law.com

With a copy to:

SmithAmundsen LLC 201 North Illinois Street, Suite 1400 Capital Center, South Tower Indianapolis, Indiana 46204-4212 Attn: John W. Tanselle

FAX: (317) 464-4149 Email: jtanselle@salawus.com

or to such substituted address as any of them have given to the other in writing.

- 13.4 <u>Headings</u>. The headings in this Agreement have been inserted solely for the ease of reference and should not be considered in the interpretation or construction of this Agreement.
- 13.5 <u>Severability</u>. In case any one or more of the provisions contained herein shall, for any reason, be held to be invalid, illegal, or unenforceable in any respect, such invalidity, illegality, or unenforceability shall not affect any other provision of this Agreement, but this Agreement shall be construed as if such invalid, illegal, or unenforceable provision or provisions had never been contained herein.
- 13.6 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be an original, but such counterparts shall together constitute one and the same instrument. In addition, this Agreement and the documents to be delivered hereunder may be executed by the parties hereto either manually or by electronic or facsimile signatures, each of which shall constitute an original signature.
- 13.7 <u>Governing Law</u>. This Agreement is executed in and shall be construed in accordance with the laws of the State of Ohio, without regard to choice of law principles.
- 13.8 Entire Agreement. This Agreement supersedes any other agreement, whether oral or written, between F&M and LBI relating to the matters contemplated hereby, and constitutes the entire agreement between the parties hereto.
- 13.9 <u>Expenses</u>. F&M and LBI shall each pay their own expenses incidental to the transactions contemplated hereby. It is understood that the fees of the investment bankers for the fairness opinion desired hereunder shall be borne by the engaging party whether or not the Merger is consummated. This provision shall survive the Effective Date or the earlier termination of this Agreement.
- 13.10 <u>Securityholder Litigation</u>. Each party shall notify the other party hereto in writing of any litigation related to this Agreement, the Merger or the other transactions contemplated by this Agreement that is brought, or, to the knowledge of either party, threatened in writing, against it and/or the executive officers or members of its Board of Directors (a "**Transaction Litigation**"), and shall keep the other party reasonably informed with respect to the status thereof. Each party shall give the other party the opportunity to participate in the defense or settlement of any Transaction Litigation, and, except to the extent required by applicable law, neither party shall settle, agree to any undertakings or approve or otherwise agree to any waiver that may be sought in connection with such Transaction Litigation, without the prior written consent of the other party (which shall not be unreasonably withheld, conditioned or delayed).

- 13.11 <u>Certain Definitions</u>. For purposes of this Agreement, "**LBI's Management**" means either of Andrew J. Briggs or Phillip Lucas; and "**F&M's Management**" means either of Paul S. Siebenmorgen or Barbara J. Britenriker. The phrases "to the knowledge of", "known to" and similar formulations with respect to LBI's Management or F&M Management means matters that are within the actual conscious knowledge of such persons after due inquiry. For purposes of this Agreement, "**business day**" means any day other than a Saturday, Sunday or other day that a federal savings bank or a national banking association is authorized or required by applicable law to be closed.
- 13.12 <u>Survival of Contents</u>. The provisions of Sections 7.8, 8.4, 10.2, 13.9 and this Section 13.12 shall survive beyond the termination of this Agreement. The provisions of Sections 7.8, 8.2, 8.4, 8.5, 13.9 and this Section 13.12 shall survive beyond the Effective Date.

[THIS SPACE INTENTIONALLY LEFT BLANK.]

IN WITNESS WHEREOF, F&M and LBI have made and entered into this Agreement as of the day and year first above written and have caused this Agreement to be executed and attested by their duly authorized officers.

FARMERS & MERCHANTS BANCORP, INC.

By: /s/ Paul S. Siebenmorgen

Paul S. Siebenmorgen, President and Chief Executive Officer

LIMBERLOST BANCSHARES, INC.

By: /s/ Andrew J. Briggs

Andrew J. Briggs, Chairman

EXHIBIT A

BANK MERGER AGREEMENT

EXHIBIT A

AGREEMENT AND PLAN OF MERGER

Merging

BANK OF GENEVA,

an Indiana commercial bank,

with and into

FARMERS & MERCHANTS STATE BANK

an Ohio commercial bank

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement and Plan"), made and entered into as of the 17th day of August, 2018, by and between **FARMERS & MERCHANTS STATE BANK**, an Ohio commercial bank ("F&M Bank"), and **BANK OF GENEVA**, an Indiana commercial bank (the "Bank") (F&M Bank and the Bank are sometimes referred to collectively as the "Constituent Companies").

WITNESSETH

WHEREAS, the Constituent Companies desire to consummate the business combination transaction outlined in this Agreement and Plan pursuant to which the Bank will merge with and into F&M Bank in accordance with 12 U.S.C. §1828(c), Chapter 11 of the Ohio Revised Code and the Indiana Financial Institutions Act (collectively, the "Law");

WHEREAS, this Agreement and Plan is being executed in connection with, and the consummation of this Agreement and Plan is expressly contingent upon the closing of, that certain Agreement and Plan of Reorganization and Merger (the "Merger Agreement") between Farmers & Merchants Bancorp, Inc., an Ohio corporation ("F&M") and Limberlost Bancshares, Inc., an Indiana corporation ("LBI") dated as of even date herewith (the "Holding Company Merger");

WHEREAS, the Boards of Directors of both F&M Bank and the Bank have approved the transactions contemplated by this Agreement;

WHEREAS, F&M, as the sole shareholder of F&M Bank, and LBI, as the sole shareholder of the Bank, have also approved the transactions contemplated by this Agreement and Plan;

NOW, THEREFORE, in consideration of the premises and of the mutual provisions, agreements, covenants, conditions and grants contained in this Agreement and Plan, and in accordance with the provisions of the Law, the parties mutually covenant and agree as follows:

ARTICLE I

THE MERGER

- 1.1 The Merger. At the "Effective Time" (as defined below), the Bank shall be merged with and into F&M Bank in accordance with applicable provisions of the Law (the "Merger"). The separate existence and company organization of the Bank shall cease, and the company existence of F&M Bank, including all its purposes, powers and objectives, shall continue unaffected and unimpaired by the Merger. F&M Bank shall continue to be governed by the laws applicable to state-chartered nonmember commercial banks under the laws of the State of Ohio and the regulations promulgated thereunder and shall succeed to all the rights, privileges, immunities, powers, duties and liabilities of the Bank as set forth in the Law.
- 1.2 **Further Assurances.** If, after the Effective Time, F&M Bank shall consider or be advised that any further deeds, assignments or assurances in the Law or any other things are necessary or desirable to (a) vest, perfect or confirm, of record or otherwise, in F&M Bank, its right, title or interest in, to or under any rights, properties or assets of the Bank, or (b) otherwise carry out the purposes of this Agreement and Plan, the Bank and its officers and directors shall be deemed to have granted to F&M Bank an irrevocable power of attorney to execute and deliver all such deeds, assignments or assurances in law and to do all acts necessary or proper to vest, perfect or confirm title to and possession of such rights, properties or assets in F&M Bank and otherwise to carry out the purposes of this Agreement and Plan, and the officers and directors of F&M Bank are authorized in the name of the Bank or otherwise to take any and all such action.
- 1.3 Offices. Immediately following the Merger, F&M Bank's principal office shall be located at 307 N. Defiance Street, Archbold, Ohio 43502 and the Bank's offices shall become branch offices of F&M Bank.
- 1.4 <u>Trust Operations</u>. In connection with the Merger, F&M Bank will apply to the ODFI and FDIC for trust powers so as to be able to continue to deliver after the Merger the fiduciary services currently provided by the Trust Department of the Bank. The Bank and F&M Bank shall cooperate in regard to the servicing of such fiduciary relationships, which, as a result of the Merger, will be transferred to F&M Bank. In the event that F&M Bank is unable to acquire trust powers or the parties otherwise agree, F&M Bank and the Bank shall work together with the Bank's customers to transfer the fiduciary relationships in an orderly fashion to a third party provider of fiduciary services.

ARTICLE II

ARTICLES OF INCORPORATION, CODE OF REGULATIONS, BOARD OF DIRECTORS AND OFFICERS

- 2.1 Name. The name of the surviving bank shall be "Farmers & Merchants State Bank."
- 2.2 **Articles of Incorporation.** The Articles of Incorporation of F&M Bank shall be the Articles of Incorporation of the surviving bank,

- 2.3 <u>Code of Regulations</u>. The Code of Regulations of F&M Bank (the "Code of Regulations") shall be the Code of Regulations of the surviving bank.
- 2.4 Officers and Directors. The directors of F&M Bank shall all remain directors of the surviving bank and shall hold such offices from the Effective Time until their respective successors are duly elected and qualified in the manner provided in the Code of Regulations; provided, Andrew J. Briggs shall be appointed to the Board of Directors of F&M Bank as contemplated by Section 8.12 of the Merger Agreement. The officers of F&M Bank shall all remain officers of the surviving bank and shall hold such offices from the Effective Time until their respective successors are duly elected and qualified in the manner provided in the Code of Regulations.

ARTICLE III

CAPITAL STOCK OF THE SURVIVING BANK

- 3.1 Shares of the Bank. At the Effective Time, by virtue of the Merger and without any further action on the part of F&M Bank or the Bank, all two hundred seventy-five thousand (275,000) issued and outstanding shares of the common stock of the Bank, whose separate existence shall cease, shall automatically and by operation of law be canceled, void and of no further effect.
- 3.2 **Shares of F&M Bank.** At the Effective Time, by virtue of the Merger and without any further action on the part of F&M Bank or the Bank, all two thousand six hundred (2,600) issued and outstanding shares of the common stock of F&M Bank, shall represent all of the issued and outstanding shares of the common stock of the surviving bank.

ARTICLE IV

SHAREHOLDER APPROVAL

F&M, as the sole shareholder of F&M Bank, and LBI, as the sole shareholder of the Bank, have approved and consented to this Merger.

ARTICLE V

GENERAL PROVISIONS

- 5.1 <u>Condition Precedent to Closing.</u> The following conditions must be satisfied prior to the closing of the Merger:
- (a) appropriate approvals must be obtained from or notices filed with the Ohio Division of Financial Institutions, Indiana Department of Financial Institutions, Board of Governors of the Federal Reserve and the Federal Deposit Insurance Corporation; and
 - (b) the Holding Company Merger must occur.

- 5.2 **Effective Time.** The Merger shall become effective immediately following the Holding Company Merger, or at such later time as designated by F&M and otherwise approved by the Ohio Division of Financial Institutions and the Federal Deposit Insurance Corporation, on the date and at the time specified in the Certificate of Merger as filed with the Secretary of State of the State of Ohio and the Articles of Merger as filed with the Secretary of State of the State of Indiana providing for the merger of the Bank with and into F&M Bank (the "Effective Time").
- 5.3 **Manner of Termination.** This Agreement and Plan and the transactions contemplated hereby may be terminated at any time prior to the Effective Time:
 - (a) by the mutual consent of F&M Bank and the Bank; or
 - (b) automatically and without further action by either F&M Bank or the Bank if the Merger Agreement is terminated for any reason.
- 5.4 **Effect of Termination.** Upon termination as provided in Section 5.3, this Agreement and Plan shall be void and of no further force or effect, and there shall be no obligation on the part of F&M Bank or the Bank or their respective officers, directors, employees, agents, or shareholders, except for payment of their respective expenses in connection with this Agreement and Plan.

[The remainder of this page was intentionally left blank.]

#F&M BANK"

FARMERS & MERCHANTS STATE BANK, an Ohio State commercial bank,

By:
Printed: Paul S. Siebenmorgen
Its: President and Chief Executive Officer

ATTEST:

Secretary/Cashier

BANK OF GENEVA, an Indiana State commercial bank,

By:
Printed: Andrew J. Briggs
Its: President

IN WITNESS WHEREOF, the parties have executed this Agreement and Plan by their respective officers duly authorized as of the date and year

first above written.

Secretary/Cashier

EXHIBIT B

VOTING AGREEMENT

EXHIBIT B

VOTING AGREEMENT

THIS VOTING AGREEMENT (this "Agreement") is entered into this 17th day of August, 2018, by and among FARMERS & MERCHANTS BANCORP, INC., an Ohio corporation ("F&M"), and the undersigned shareholders or option holders (each, a "Shareholder", and collectively, the "Shareholders") of LIMBERLOST BANCSHARES, INC., an Indiana corporation ("LBI").

WITNESSETH:

In consideration of the execution by F&M of the Agreement and Plan of Reorganization and Merger between F&M and LBI of even date herewith (the "Merger Agreement"), the undersigned Shareholders of LBI hereby agree that each of them shall cause all LBI common shares owned by him/her of record and beneficially, including, without limitation, all shares owned by him/her individually, all shares owned jointly by him/her and his/her spouse, all shares owned by any minor children (or any trust for their benefit), all shares owned by any business of which any of the Shareholders who are directors are the principal shareholders (but in each such case only to the extent the Shareholder has the right to vote or direct the voting of such shares), and specifically including all shares shown as owned directly or beneficially by each of them on Schedule 1 attached hereto or acquired subsequently hereto (collectively, the "Shares"), to be voted in favor of the merger of LBI with and into F&M in accordance with and pursuant to the terms of the Merger Agreement at the annual or special meeting of shareholders of LBI called for that purpose. Notwithstanding any other provision of this Agreement to the contrary, each Shareholder shall be permitted to vote such Shares in favor of another Acquisition Proposal (as such term is defined in the Merger Agreement) that is submitted for approval by the shareholders of LBI if both of the following shall have occurred: (a) LBI's Board of Directors has approved such Acquisition Proposal and recommended such Acquisition Proposal to LBI's shareholders in accordance with Section 7.5 of the Merger Agreement and (b) the Merger Agreement has been terminated in accordance with Section 10.1(f) thereof.

Each of the Shareholders further agrees and covenants that he/she shall not sell, assign, transfer, dispose or otherwise convey, nor shall he/she cause, permit, authorize or approve the sale, assignment, transfer, disposition or other conveyance of, any of the Shares or any interest in the Shares to any other person, trust or entity (other than LBI) prior to the annual or special meeting of shareholders of LBI called for the purpose of voting on the Merger Agreement without the prior written consent of F&M, such consent not to be unreasonably withheld in the case of a gift or similar estate planning transaction (it being understood that F&M may decline to consent to any such transfer if the person acquiring such Shares does not agree to take such Shares subject to the terms of this Agreement).

This Agreement shall be governed by and construed in accordance with the laws of the State of Ohio, without regard to conflict of laws provisions thereof. This Agreement may be executed in counterparts, each of which (including any facsimile or Adobe PDF copy thereof) shall be deemed to be an original, but all of which shall constitute one and the same agreement. It is understood and agreed that Shareholders who execute this Agreement shall be bound hereby, irrespective of whether all Shareholders execute this Agreement. The obligations of each of the Shareholders under the terms of this Agreement shall terminate contemporaneously with the termination of the Merger Agreement.

Notwithstanding any other provision hereof, nothing in this Agreement shall be construed to prohibit a Shareholder, or any officer or affiliate of a Shareholder who is or has been designated a member of LBI's Board of Directors, from taking any action solely in his or her capacity as a member of LBI's Board of Directors or from exercising his or her fiduciary duties as a member of LBI's Board of Directors to the extent specifically permitted by the Merger Agreement.

[Signatures appear on following pages.]

IN WITNESS WHEREOF, F&M and each of the undersigned Shareholders of LBI have made and executed this Agreement as of the day and year first above written.

FARMERS	Q٠	MFR	CHA	2TM	BA	NCORP	INC
L'AINMEINO	œ	IVILIA		כנוגר	$D\Lambda$	MODIAL.	IIIV.

By:			
	ъ	10	0:1

Paul S. Siebenmorgen, President and Chief Executive Officer

SHAREHOLDERS	
-	
_	
-	
-	
 -	

SCHEDULE 1

LISTING OF SHARES

SECTION 1 Name	SECTION 2 Amount

Exhibit 99.1

Company Press Release dated August 20, 2018

NEWS RELEASE AUGUST 20, 2018

Farmers & Merchants Bancorp, Inc. 307 N. Defiance Street Archbold, Ohio 43502

Farmers & Merchants Bancorp, Inc. Contact: Marty Filogamo Senior Vice President – Marketing Manager Farmers & Merchants Bancorp, Inc. (419) 445-3501 ext. 15435 mfilogamo@fm.bank

Limberlost Bancshares, Inc. Contact:

Andrew J. Briggs Chairman Limberlost Bancshares, Inc. (260) 368-7288 ABriggs@bankofgeneva.com

For Release: Immediately

FARMERS & MERCHANTS BANCORP, INC. AGREES TO ACQUIRE LIMBERLOST BANCSHARES, INC.

ARCHBOLD, Ohio (BUSINESS WIRE) Farmers & Merchants Bancorp, Inc. ("F&M") (Nasdaq-Capital Market: FMAO), the holding company for Farmers & Merchants State Bank, announced today that they have signed a definitive merger agreement whereby F&M will acquire Limberlost Bancshares, Inc. ("Limberlost"), Geneva, Indiana, the holding company for Bank of Geneva, in a combination stock and cash transaction. Bank of Geneva operates six full-service offices in northeast Indiana and has approximately \$287 million in assets, \$257 million in loans, \$212 million in deposits and \$30.6 million in consolidated equity as of June 30, 2018. Limberlost is taxed as an S-Corporation.

Subject to the terms of the merger agreement, which has been approved by the Board of Directors of each company, Limberlost shareholders will receive 1,830 shares of F&M common stock for each outstanding share of Limberlost common stock plus \$8,465 in cash. Limberlost currently has 1,000 shares of common stock outstanding. Based on the F&M closing share price as of August 17, 2018, the transaction is valued at approximately \$88.765 million in aggregate.

The transaction will significantly increase F&M's presence in northeast Indiana. Bank of Geneva operates offices in the Indiana communities of Geneva, Berne, Monroe, Monroeville, Portland and Decatur. F&M has 21 offices in Ohio and 4 offices in Indiana.

Paul Siebenmorgen, President and CEO of F&M, stated "F&M is pleased to combine our two strong institutions, which have a shared vision and shared values. This is an excellent opportunity for the Bank of Geneva to become part of a larger community banking organization that will allow us to offer customers in the communities served a wider range of financial services while still preserving the convenience of local decision makers throughout our office network."

Following the merger, Andrew Briggs, Chairman of Limberlost, will be appointed to the board of directors of F&M and Farmers & Merchants State Bank. In addition, Phil Lucas, currently Executive Vice President of Bank of Geneva, will join Farmers & Merchants State Bank as Senior Vice President.

Andrew Briggs stated, "We are very excited to be joining forces with another community bank who is larger than the Bank of Geneva. We share the same commitment to our customers and the communities that we serve. A great benefit to our customers is that decisions remain at the local level and they will benefit from a wider range of products and services. We believe that our success has been based on our customers' trust in us and the great employees that we have had the pleasure of working with over the last 125 years."

Excluding one-time transaction costs, F&M expects the transaction to be approximately 12.8% and 15.5% accretive to first and second-year fully diluted earnings per share, respectively. Tangible book value per share will be diluted approximately 6.1% at closing including the impact of an estimated \$5.5 million of combined pre-tax transaction costs. The tangible book value dilution is expected to be recovered in less than three years using the crossover method.

When the transaction is completed, F&M will maintain consolidated assets of approximately \$1.5 billion. Subject to customary regulatory approvals, Limberlost shareholder approval and other conditions set forth in the definitive merger agreement, the transaction is anticipated to close in the fourth quarter of 2018 or first quarter of 2019.

F&M is being advised by ProBank Austin and Shumaker, Loop & Kendrick, LLP. Limberlost is being advised by Renninger & Associates, LLC and SmithAmundsen LLC.

Important Information for Investors and Shareholders:

This press release does not constitute an offer to sell or the solicitation of an offer to buy securities of F&M. F&M will file a registration statement on Form S-4 and other documents regarding the proposed business combination transaction referenced in this press release with the Securities and Exchange Commission ("SEC") to register the shares of F&M's common stock to be issued to the shareholders of Limberlost. The registration statement will include a proxy statement/prospectus which will be sent to the shareholders of Limberlost in advance of a special meeting of Limberlost shareholders that will be held to consider the proposed merger. Investors and Limberlost shareholders are urged to read the proxy statement/prospectus and any other relevant documents to be filed with the SEC in connection with the proposed transaction because they will contain important information about F&M, Limberlost and the proposed transaction. Investors and shareholders may obtain a free copy of these documents (when available) through the website maintained by the SEC at www.sec.gov. These documents may also be obtained, without charge, by directing a request to Farmers & Merchants Bancorp, Inc., 307 N. Defiance Street, Archbold, Ohio 43502: Attention Investor Relations.

Limberlost and its directors and executive officers may be deemed to be participants in the solicitation of proxies from the shareholders of Limberlost in connection with the proposed merger. Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the proxy statement/prospectus regarding the proposed merger when it becomes available. Free copies of this document may be obtained as described in the preceding paragraph, when it becomes available.

About Farmers & Merchants Bancorp, Inc.:

Farmers & Merchants Bancorp, Inc., Archbold, Ohio, is the holding company of Farmers & Merchants State Bank (the "Bank"), which was established in 1897. The Bank serves the financial needs of individuals, farmers, businesses, and industries by offering traditional banking products, as well as online, mobile and telephone banking products. F&M's common shares are traded on the NASDAQ Capital Market exchange under the symbol "FMAO." Additional information on F&M may be found on its website: www.fm.bank.com.

About Limberlost Bancshares, Inc.:

Limberlost Bancshares, Inc., Geneva, Indiana, is the holding company of Bank of Geneva, which was established in 1892. Additional information about Limberlost and Bank of Geneva may be found on its website: www.bankofgeneva.com.

Safe Harbor Statement:

Statements made in this news release that are not historical facts are "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, Section 21E of the Securities Exchange Act of 1934, as amended, and the Private Securities Litigation Reform Act of 1995. These statements are subject to certain risks and uncertainties including, but not limited to, the successful completion and integration of the transaction contemplated in this release, which includes the retention of the acquired customer relationships, adverse changes in economic conditions, the impact of competitive products and pricing and the other risks set forth in the F&M's filings with the SEC. As a result, actual results may differ materially from the forward-looking statements in this news release.

F&M encourages readers of this news release to understand forward-looking statements to be strategic objectives rather than absolute targets of future performance. F&M undertakes no obligation to update these forward-looking statements to reflect events or circumstances after the date of this news release or to reflect the occurrence of unanticipated events, except as required by applicable legal requirements. Copies of documents filed with the SEC are available free of charge at the SEC's website at http://www.sec.gov and/or from F&M's website.